### SECURITIES AND EXCHANGE COMMISSION

## FORM 8-K

Current report filing

Filing Date: **2000-03-28** | Period of Report: **2000-03-15** SEC Accession No. 0000950136-00-000419

(HTML Version on secdatabase.com)

## **FILER**

#### **OFFICIAL INFORMATION CO**

CIK:847841| IRS No.: 731341805 | State of Incorp.:DE | Fiscal Year End: 1231

Type: **8-K** | Act: **34** | File No.: **033-57587** | Film No.: **580660** SIC: **2721** Periodicals: publishing or publishing & printing

Mailing Address 250 WEST 57TH STREET NEW YORK NY 10019 Business Address 250 WEST 57TH STREET SUITE 2421 NEW YORK NY 10019 2122475160

# SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 8-K

#### CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

DATE OF REPORT (DATE OF EARLIEST EVENT REPORTED)

MARCH 15, 2000

THE OFFICIAL INFORMATION COMPANY (EXACT NAME OF REGISTRANT AS SPECIFIED IN ITS CHARTER)

DELAWARE 1-10263 73-1341805 (STATE OR OTHER JURISDICTION OF (COMMISSION (I.R.S. EMPLOYER INCORPORATION OR ORGANIZATION) FILE NUMBER) IDENTIFICATION NO.)

250 West 57th Street, Suite 2421, New York, New York 10019
(ADDRESS OF PRINCIPAL EXECUTIVE OFFICES) (ZIP CODE)

REGISTRANT'S TELEPHONE NUMBER, INCLUDING AREA CODE: (212) 247-5160

#### ITEM 2. ACQUISITION OR DISPOSITION OF ASSETS

As of March 15, 2000, The Official Information Company (the "Company"), through a subsidiary, acquired in a series of integrated transactions (a) all of the capital stock of (i) Business Mutual Association of Mass., Inc., a Massachusetts corporation, (ii) Stores Protective Association, Inc., a California corporation, (iii) Employers Screening Service, Inc., an Alabama corporation, and (iv) Stores Mutual Association of Illinois, Inc., an Illinois corporation pursuant to a Stock Acquisition Agreement dated as of March 15, 2000 and (b) all of the assets of (i) United States Mutual Association, Inc., an Illinois corporation, (ii) Employers Mutual Association West (a division of Prevention Technology Associates, Inc.), (iii) Executive Management Services, Inc. (dba Employers Mutual Association), a Pennsylvania corporation, (iv) PenJerDel Mutual Association, a Delaware corporation, (v) National Loss Prevention Bureau Group, Inc. and National Loss Prevention Bureau Group, Ltd., New York corporations, and (vi) SAMIAM, LLC (dba Upper Midwest Mutual Association), a Minnesota LLC, pursuant to an Asset Acquisition Agreement dated as of March 15, 2000. The aggregate consideration paid by the Company to the transferors was \$8.3 million in the form of stock of its subsidiary, TISI Holdings, Inc., and \$15.2 million in cash (subject to adjustment for working capital). The Company financed the cash portion of the acquisition from its \$25 million Credit Agreement with First Union National Bank.

#### (c) Exhibits

- 2.1 Form of Stock Acquisition Agreement dated as of March 15, 2000 among TISI Holdings, Inc., The Official Information Company and the transferor(s).
- 2.2 Form of Asset Acquisition Agreement dated as of March 15, 2000 among TISI Holdings, Inc., The Official Information Company and the transferor(s).
- 4. Stockholders Agreement dated as of March 15, 2000 among TISI Holdings, Inc., The Official Information Company and the other signatories thereto.

#### SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Date: March 24, 2000

THE OFFICIAL INFORMATION COMPANY

By /s/ Ian L.M. Thomas

\_\_\_\_\_

IAN L.M. THOMAS
PRESIDENT AND CHIEF EXECUTIVE OFFICER

#### STOCK ACQUISITION AGREEMENT

March , 2000

The parties to this agreement are the parties listed on exhibit I (collectively, the "Stockholders"), TISI Holdings, Inc., a Delaware corporation ("TISH"), which is a wholly-owned subsidiary of TOIC (hereafter defined), and The Official Information Company, a Delaware corporation ("TOIC").

The Stockholders own shares of common stock	(constituting all
of the issued and outstanding shares of capital stock) of	
, a corporation (the "Company"	"), which
operates the businesses of pre-employment investigation and screen	eening (the
"Business"). The parties have agreed upon the transfer to TISH $\alpha$	of all of the
shares of the Company (the "Shares"). The stockholders of the as	ffiliates of the
Company, and in certain instances the affiliates themselves, lis	sted on exhibit
II (the "Affiliates") have entered into acquisition agreements w	with TISH on the
date hereof (the "Affiliate Acquisition Agreements") whereby eit	ther the assets
or stock of each of the Affiliates will be transferred to TISH of	on the terms set
forth in the respective Affiliate Acquisition Agreements.	

TOIC owns all of the outstanding shares of capital stock of Total Information Services, Inc., an Oklahoma corporation ("TISI"), which, among other things, conducts a business similar to the Business. In connection with the transactions contemplated by this agreement, TOIC is transferring and contributing all of such shares of TISI to TISH.

Accordingly, it is agreed as follows:

- 1. Transfer of Shares.
- 1.1 Transfer at Closing. At the closing referred to in section 3, the Stockholders shall transfer to TISH, and TISH shall acquire from the Stockholders, the Shares, free and clear of all Liens (as defined herein).
- 1.2 Pre-Closing Distribution. Immediately prior to the closing the Company shall assign and transfer to the Stockholders and the Stockholders shall accept from the Company all the shares of capital stock held by the Company in United States Mutual Association, Inc. [and all the assets (the "Retained Assets") and liabilities relating to the refund checking and civil recovery databases of the Company, as set forth on schedule 1.2 (the "Excluded Business").] The distribution[s] described in this section 1.2 shall be referred to as the "Pre-Closing Distribution".
  - 2. Purchase Price.

#### 2.1 Amount of Consideration. As full consideration for the Shares:

(a) at the closing, TISH shall pay to the Stockholders, by
wire transfer of immediately available funds, the sum of \$, subject to
adjustment as provided in section 2.3, which shall be payable as follows: (i)
\$ to the Stockholders, (ii) \$ into a trust account established by
the Stockholders and the Affiliates pursuant to an agreement of even date
herewith, as more fully described on exhibit III, and (iii) \$ to the escrow
agent under an escrow agreement in the form of exhibit IV ("Escrow Agreement")

(b) at the closing, TISH shall cause to be issued to the Stockholders \_\_\_\_ shares of Class B Non-Voting Common Stock of TISH and \_\_\_ shares of Class C Non-Voting Common Stock of TISH, as more fully described on exhibit V (collectively, the "TISH Shares");

2.2 Distributions. The Stockholders acknowledge that they have instructed TISH to make the cash payments and to cause the TISH Shares to be issued in the manner provided in section 2.1 and TISH bears no responsibility with respect to the further distribution of such payments or TISH Shares.

#### 2.3 Working Capital Minimum.

(a) Within 90 days of the Closing Date, TISH shall cause to be delivered to the Stockholders a statement of assets and liabilities of the Company (the "Closing Balance Sheet") prepared on an accrual basis in accordance with generally accepted accounting principles as of the close of business on the Closing Date. If the sum of the current liabilities of the Company shown on the Closing Balance Sheet is greater than the sum of the current assets of the Company shown on the Closing Balance Sheet, the Stockholders shall pay in cash to TISH the amount that the sum of the current liabilities exceeds the sum of the current assets (the "Adjustment Amount"), with such payment to be made within 30 days of the final determination of the Adjustment Amount. In the event that the sum of the current assets of the Company shown on the Closing Balance Sheet is greater than the sum of the current liabilities of the Company shown on the Closing Balance Sheet, TISH shall pay in cash to the Stockholders the amount that the sum of the current assets exceeds the sum of the current liabilities (the "Reverse Adjustment Amount") with such payment to be made within 30 days of the final determination of the Reverse Adjustment Amount.

(b) Within 30 days after the Stockholders' receipt of the Closing Balance Sheet, the Stockholders may deliver to TISH a written statement setting forth any disagreements of the Stockholders with any of the information contained in the Closing Balance Sheet that affects the determination of the Adjustment Amount (or the Reverse Adjustment Amount). If the Stockholders do not submit such a statement within the 30-day period, then the Adjustment Amount (or the Reverse Adjustment Amount), if any, as determined in accordance with the Closing Balance Sheet, shall be final and binding on the parties. If TISH

and the Stockholders do not resolve the disagreements within 15 days after delivery of the written statement by the Stockholders, a firm of independent certified public accountants, whose decisions on the matter shall be binding and whose fees and expenses shall be borne 50% by the Stockholders and 50% by TISH, shall be designated by agreement between them; if they fail to agree on the firm to decide the matter within an additional 10 days, the accountants shall be selected by the president of the American Institute of Certified Public Accountants. Each party shall allow the other party reasonable access to all information reasonably requested by such other party to prepare the Closing Balance Sheet or the Stockholders' statement hereunder, as the case may be.

- 3. Closing. The closing under this agreement shall take place at the offices of Proskauer Rose LLP, 1585 Broadway, New York, New York 10036, on March 15, 2000. The date on which the closing is held is referred to in this agreement as the "Closing Date." At the closing, the parties shall execute and deliver the documents referred to in section 9. If the closing has not occurred on or prior to June 30, 2000, either party may terminate this agreement by notice to the other; upon such termination, neither party shall have any liability arising out of this agreement other than any liability resulting from its breach of or misrepresentation under this agreement prior to termination.
- 4. Satisfaction of Liabilities; Cancellation of Affiliate Payables and Receivables.
- (a) Prior to the closing, the Stockholders shall cause the Company to satisfy all of its liabilities, contingent or otherwise, except those disclosed on schedule 4 (the "Permitted Liabilities"). [Schedule 4 will define Permitted Liabilities in the same manner as Assumed Liabilities are defined in the Asset Acquisition Agreement.]
- (b) Immediately prior to the closing, any payable or other liability to, or receivable or other asset due from, a Stockholder, any affiliate of a Stockholder, or any Affiliate will be canceled without any cost or adverse consequence to the Company.
- 5. Representations and Warranties of the Stockholders. The Stockholders jointly and severally represent and warrant to TISH and TOIC that, as of the date hereof:
- 5.1 Organization and Authority. The Company is a corporation duly organized, validly existing and in good standing under the law of the State of \_\_\_\_\_\_ and has the full corporate power and authority to carry on its business as it currently is being conducted. The Company is duly qualified to do business and is in good standing as a foreign corporation in each jurisdiction listed on schedule 5.1, which is each jurisdiction in which it owns, leases or operates property or is otherwise required to be qualified, except where the failure to be so qualified would not have a material adverse effect on the Business. The Company does not have any equity or other similar interests in any entity (other than United States Mutual Association, Inc.). The Stockholders have the full legal right, power and authority to enter into and perform this

employment agreements referred to in sections 2.1(a)(iii), 7.8 and 7.9 (the "Closing Documents") and no spousal consent is required.

- 5.2 Authorization of Agreement. This agreement constitutes, and the Closing Documents when executed at the closing will constitute, the valid and binding obligation of each Stockholder and is enforceable against each Stockholder in accordance with its respective terms, except as enforceability may be limited by bankruptcy, insolvency or other similar laws affecting the enforcement of creditors' rights in general and subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- 5.3 Consents of Third Parties. Subject to receipt of the consents and approvals referred to on schedule 5.3, the execution, delivery and performance of this agreement and Closing Documents by each Stockholder will not (i) conflict with the certificate of incorporation or bylaws of the Company; (ii) conflict with, or result in the breach or termination of, or constitute a default under, any agreement, lease, commitment, order or other instrument (including, without limitation, those listed on schedule 5.10), or any order, judgment or decree, to which any Stockholder is a party or by which any Stockholder is bound; (iii) constitute a violation by any Stockholder of any law or regulation applicable to him or her; or (iv) result in the creation of any lien, charge, claim or encumbrance ("Lien") upon any of the capital stock to be sold to TISH under this agreement or any of the properties or assets of the Company. Except as set forth on schedule 5.3, no consent, approval or authorization of, or designation, declaration or filing with, any governmental authority is required on the part of any Stockholder or the Company in connection with the execution, delivery and performance of this agreement. In this regard each Stockholder has executed and delivered to TISH the affidavit attached to this agreement as Exhibit VI.
- consists of \_\_\_\_ shares of common stock, par value \$\_\_\_ per share, of which \_\_\_ shares are issued and outstanding as of the date of this agreement. All the Company's outstanding shares were duly authorized for issuance, were validly issued, and are fully paid and nonassessable and are owned by the Stockholders. There are no outstanding options, warrants or rights of any kind to acquire any shares or interests of any class of security, and there are no outstanding securities convertible into any shares or interests of any class of security, of the Company, nor are there any obligations to issue any such options, rights or securities. There are no existing shareholders agreements, voting agreements or any other existing arrangements that require or permit any shares or interests of the Company to be voted by or at the discretion of anyone other than the record owner and there are no restrictions of any kind on the transfer of any of those shares or interests, except those imposed by applicable United States federal and state securities laws.

5.5 Ownership of Shares. The Stockholders are the record and beneficial owners of an aggregate of \_\_\_\_ shares of common stock of the Company, as provided on schedule 5.5, free and clear of all Liens, and those shares represent all of the outstanding capital stock of the Company. At the closing, TISH will receive valid title to all of the outstanding shares of the Company free and clear of all Liens.

- 5.6 Title to Assets and Related Matters; Condition of Assets.
- (a) The Company has valid title to all of the assets (tangible and intangible) used in the Business or presently located on its premises ([except for the Retained Assets,] the "Assets") free and clear of any Lien, except for the lien, if any, of current taxes not yet due and payable and any other lien disclosed on schedule 5.6(a). The Assets include all of the assets reflected on the balance sheet as of December 31, 1999 referred to in section 5.7, except for the [the Retained Assets and] items sold, transferred or otherwise disposed of since that balance sheet date in the ordinary course of business and substantially consistent with past practice.
- (b) The Assets include all the assets (tangible and intangible) necessary for, or used in, the conduct of the Business in the manner in which it has been conducted by the Company [(except for the Retained Assets)] and those assets (together with replacements for the excluded items) will be sufficient to enable the Company to continue to conduct the Business after the closing in the manner in which it has been conducted by the Company previously.
- (c) Schedule 5.6(c) contains a list of all real properties owned or leased by the Company with respect to the Business. The Company enjoys peaceful and undisturbed possession under all real property leases covering premises used in the conduct of the Business, and all such real property leases are valid and subsisting and in full force and effect. Except as set forth on schedule 5.6(c), all improvements on the real properties used by the Company in connection with the Business are in accordance, in all material respects, with all applicable laws, ordinances, regulations and orders, including, but not limited to, those applicable to zoning, environment and the establishment and maintenance of working conditions for labor. All of the buildings and structures leased to or owned by the Company in connection with the Business are, in all material respects, in good condition and are adequate and suitable for their present uses and purposes, and, except as otherwise disclosed on schedule 5.6(c), the transactions contemplated by this agreement will not adversely affect the right of TISH to use those properties for the same purpose and to the same extent after the Closing Date as they were being used by the Stockholders prior to the date of this agreement. To the knowledge of the Stockholders or the Company, there are no interior or exterior material structural defects or other material defects in the improvements or any material defect in the plumbing, electrical, mechanical, heating, ventilating or air-conditioning systems or other systems. To the knowledge of the Stockholders or the Company, all such systems are, in all material respects, in good working order, and all roofs and basements are, in all material respects, in good condition and free of leaks.

With respect to the real property owned by the Company, (i) the Stockholders have delivered to TISH a copy of any title insurance policies and surveys for such property in the Stockholders' possession and (ii) the Company has good and marketable fee simple title thereto and owns all improvements (including buildings and other structures) thereon, subject only Liens set forth on schedule 5.6(c) and such other Liens as do not materially interfere with the operation of the real property or adversely affect the marketability or value of

the real property. To the knowledge of the Stockholders or the Company, except as disclosed on schedule 5.6(c), there are no: (A) pending or contemplated annexation, condemnation or similar proceedings affecting, or which may affect, the real property, (B) proposed or pending proceedings to change or redefine the zoning of the real property, (C) proposed changes in roads which may adversely affect ingress to or egress from the real property, or (D) proposed or pending special assessments affecting the real property or any penalties or interest due with respect to real estate taxes assessed against the real property that are payable by the Company as all taxes applicable to the real property have been paid in full, except taxes for the current year, which are not yet due and payable.

- (d) Schedule 5.6(d) contains a complete list of all equipment (including computers and office equipment), furniture, vehicles, and other tangible assets used in the Business that had an original cost for any individual item of more than \$1,000.
- (e) To the knowledge of the Stockholders or the Company, all of the Assets are, in all material respects, in good operating condition and in good condition of maintenance and repair, subject to normal wear and tear, and are suitable for continued use in the normal course of business.
- 5.7 Financial Statements. Schedule 5.7 contains financial statements consisting of (i) balance sheets of the Company as of December 31, 1997 and December 31, 1998 and the related statements of income and cash flows for the 12 month periods then ended, including all related notes thereto [with ]; and (ii) a balance sheet of the Company as of December the report of 31, 1999 and the related statements of income and cash flows for the 12 month period then ended. Except as set forth on schedule 5.7, all such financial statements have been prepared on an accrual basis or a cash basis, as indicated in such financial statements, and fairly present the financial position (on a cash or accrual basis, as applicable) of the Business as of the dates indicated and the results of operations of the Business for the periods indicated. All such financial statements have been prepared in accordance with the Company's books and records and show all income and expenses attributable to the Business during the respective periods covered by them. All of the Company's books of account relating to the Business have been exhibited or made available to TISH, and those books of account accurately record all transactions of the Business during the respective periods covered by them. All of the accounts receivable reflected on the balance sheet as of December 31, 1999 arose from, and all of

the accounts receivable as of the Closing Date will have arisen from, bona fide transactions in the ordinary course of business and, except as disclosed on schedule 5.7, none of them is or will be subject to any defense, counterclaim or setoff and all such accounts receivable shall be collectible within 90 days after such obligations arose in their full recorded amounts, subject to any reserves therefor reflected on the balance sheet as of December 31, 1999 or the Closing Balance Sheet, respectively.

5.8 Undisclosed Liabilities. Except to the extent reflected on the balance sheet as of December 31, 1999 referred to in section 5.7 or reflected in the notes to that balance sheet, or otherwise disclosed on a schedule to this agreement, the Company does not, and on the Closing Date will not, have any liability or obligation of any kind relating to the Business,

whether accrued, absolute, contingent or otherwise, other than (a) liabilities and obligations under orders, commitments, agreements and leases entered into in the ordinary course of business (which, to the extent required by section 5.10, are referred to on schedule 5.10), and (b) other liabilities and obligations that arose in the ordinary course of the Business and are not material in amount. Notwithstanding the foregoing, any liability resulting from a claim asserted by any individual against TISH after the closing with regard to a negative report issued by the Company in the ordinary course of business prior to the closing shall not be considered an undisclosed liability (but shall instead be deemed a "Permitted Liability" for all purposes of this agreement), so long as such assertion of a claim was not known by the Company to exist prior to the closing ("Unknown Report Claim").

- 5.9 Absence of Certain Changes. Since December 31, 1999, the Stockholders and the Company have operated the Business in the ordinary course and consistent with past practices, and, except as set forth on schedule 5.9:
- (a) the Company has not entered into any transaction or incurred any liability or obligation with respect to the Business that was unusual in nature or amount or was entered into or incurred other than in the ordinary course of business;
- (b) there has not been any material adverse change in the condition (financial or otherwise) or prospects of the Business, or in the Assets;
- (c) [except for the Retained Assets,] the Company has not sold or transferred any assets relating to or used in the Business other than in the ordinary course of business and other than assets that have been replaced with other assets of equal or greater value;
- (d) the Company has not (i) granted or agreed to grant any general increase in any rate or rates of salaries or compensation to employees or agents of the Business or any specific increase in the salary or compensation to any such employee or agent whose total salary and compensation after such

increase would be at an annual rate in excess of \$30,000, (ii) increased or amended any benefits of those employees or agents, (iii) amended, adopted or terminated any Plan, or (iv) entered into any employment, bonus, deferred compensation or any other agreement with any employee or agent;

- (e) the Company has not made any material change in the manner in which the Business has been operated or the accounting principles or practices employed by the Company in connection with the Business, and no person who performed substantial services with respect to the Business during the twelve months preceding the date of this agreement is currently employed by or performing substantial services for any affiliate of the Company or of the Stockholders;
- (f) the Company has not taken any action to accelerate collection of accounts receivable or to defer payment of liabilities, except in the ordinary course consistent with past practices; and
- (g) to the Stockholders' and the Company's knowledge, the Company has not amended, terminated, canceled or compromised any material claim relating to the Business, or waived or allowed to lapse or terminate any other rights of substantial value to the Business.
- 5.10 Lists of Agreements, etc. Schedule 5.10 contains, with respect to the Business, a true and complete list of: (a) all commitments and agreements for the purchase of materials or supplies or the receipt of services that involve an expenditure by the Company of more than \$10,000 for any one commitment or two or more related commitments; (b) all agreements relating to pre-employment investigation and screening services; (c) all leases or other rental agreements under which the Company is lessee or lessor; (d) all license agreements under which the Company is either licensee or licensor; (e) all collective bargaining agreements and all employment and consulting agreements and agreements with independent contractors that provide for compensation in excess of \$10,000 a year; (f) all agreements between the Company and any affiliate of the Company (or any entity in which any of them has a material interest); (q) all commitments and other agreements limiting the Company's freedom to engage in any line of business or to compete with any other person; and (h) all other agreements, commitments and orders to which the Company is a party or by which it is bound that involve more than \$10,000 or are otherwise material to the Business or the Assets. True and complete copies of the agreements, commitments and leases referred to on schedule 5.10 (if in written form and summaries thereof if in oral form) have been delivered or made available to TISH.
- 5.11 Status of Agreements. Except as set forth on schedule 5.11, all leases, commitments and other agreements of the Company were entered into in connection with and in the ordinary course of the Business and on an arm's length basis. Each of the agreements, commitments, orders and leases referred to in section 5.10 is presently in full force and effect in accordance with its terms and the Company is not in default, and, to the knowledge of the

Stockholders or the Company, no other party is in default, under any agreement, commitment, order or lease referred to in section 5.10 and each of those agreements, commitments, orders and leases is valid and binding upon and enforceable against each of the parties thereto in accordance with its terms, except as enforceability may be limited by bankruptcy, insolvency or other similar laws affecting the enforcement of creditors' rights in general. To the knowledge of the Stockholders or the Company, no party to any of the agreements, commitments, orders or leases referred to in section 5.10 has made, asserted in writing or has any defense, setoff or counterclaim under any of those agreements, commitments, orders or leases or has exercised any option granted to it to cancel or terminate its agreement, to shorten the term of its agreement, or to renew or extend the term of its agreement, and the Company has not received any written notice to that effect.

5.12 Litigation; Compliance with Laws. Except as set forth on schedule 5.12, within the past three years there has not been, and there is not currently, any claim, litigation, proceeding or governmental investigation pending or, to the knowledge of the Stockholders or the Company, threatened, or any order, injunction or decree outstanding against or relating to the Company with respect to the Business or any of the Assets. Subject to the last sentence of section 5.8, the Stockholders do not know of any basis for future claims, litigations,

proceedings or investigations against the Company with respect to the Business or any of the Assets. The Company is not in violation, in any material respect, of any law, regulation, ordinance, or any other requirement of any governmental body or court, including the Fair Credit Reporting Act, arising out of the operation of the Business (including, without limitation, the use of the databases and any information contained therein and related software) and no written notice has been received by the Company or any of its officers, directors or stockholders alleging any such violation.

5.13 Intellectual Property. Schedule 5.13 contains a true and complete list of the trademarks, trade names, copyrights, patents, trade secrets, logos, computer software, databases, web sites, domain names and e-mail addresses (collectively, "Intellectual Property"), used by the Company in the Business. Schedule 5.13 also contains a true and complete list of all patent, copyright, trademark, database and domain name registrations or registration applications that have been obtained or filed by the Company anywhere in the world. Except as set forth on schedule 5.13, the Company owns or possesses adequate licenses or other rights to use, free and clear of any Lien, each of the items of Intellectual Property listed on schedule 5.13 (including, without limitation, any information contained in the databases), and they constitute all the Intellectual Property necessary for the continued operation of the Business in a manner consistent with past practices. Schedule 5.13 identifies which items of Intellectual Property are owned by the Company and which items are used by the Company under license. Except as set forth on schedule 5.13, (a) to the knowledge of the Stockholders or the Company (i) there is no violation by others of any right of the Company with respect to its Intellectual Property and (ii)

the Company is not, in connection with the Business, infringing upon any Intellectual Property or other rights of any third party, including, without limitation, by the Company's use of the Intellectual Property; (b) no proceedings are pending or threatened, and no claim has been received by the Company or the Stockholders alleging any such violation; (c) no royalty or similar fee of any kind is payable by the Company for the use of any of the Intellectual Property; and (d) the Stockholders have not granted any person or entity any interest, as licensee or otherwise, in or to any one or more items of the Intellectual Property.

5.14 Labor Matters. Except as set forth on schedule 5.14, with respect to the Business (a) the Company is in compliance, in all material respects, with all applicable laws respecting employment and employment practices, terms and conditions of employment, and wages and hours, and is not engaged in any unfair labor practice; (b) there is no unfair labor practice charge or complaint against the Company pending before the National Labor Relations Board, any state labor relations board or any court or tribunal and, to the knowledge of the Stockholders or the Company, none is or has been threatened in writing; (c) there is no labor strike, dispute, request for representation, slowdown or stoppage actually pending against or affecting the Company and, to the knowledge of the Stockholders or the Company, none is or has been threatened in writing; (d) no grievance which might have a material adverse effect on the conduct of the Business or any arbitration proceeding arising out of or under any collective bargaining agreement is pending and, to the knowledge of the Stockholders or the Company, none is or has been threatened in writing; and (e) there is no workman compensation or similar claim pending or threatened. Except as set forth on schedule 5.14, to the knowledge of the Stockholders or the Company, there has been no sexual harassment or similar claim against any

employee of the Business and no employee of the Business has at any time within the three years preceding the date of this agreement asserted in writing against the Company or any of its employees any claim of discrimination.

- 5.15 Environmental Matters. With respect to the ownership and operation of the Business and any currently or formerly owned, operated or used real property, to the knowledge of the Stockholders, except as set forth on schedule 5.15:
- (a) The Company and all of the real property owned, leased or used by the Company (the "Real Property") is in compliance, in all material respects, with all federal, state and local laws and regulations relating to injury to or pollution or protection of human health or the environment (the "Environmental Laws"), including the common law, environmental permits and laws and regulations relating to emissions, discharges, releases or threatened releases of chemicals, pollutants, contaminants, wastes, substances, petroleum and petroleum products (collectively, "Materials of Environmental Concern"), or otherwise relating to the manufacture, generation, processing, distribution, use, treatment, storage, disposal, transport or handling of Materials of Environmental Concern;

(b) there are no past or present actions, activities, circumstances, conditions, events or incidents, including, but not limited to, the release, emission, discharge or disposal or threatened release of any Material of Environmental Concern, that could form the basis of any material claim against the Company or the Real Property or could reasonably be expected to constitute a material violation by the Company (or, after the closing, TISH) or is reasonably likely to require investigation, remediation or monitoring under the Environmental Laws, and neither the Company nor any of the Stockholders has received any written notice of actual or potential liability under the Environmental Laws; and

(c) there are no underground storage tanks located on the Real Property, there are no friable asbestos-containing materials in or on the Real Property and no polychlorinated biphenyls (PCBs) are used or stored by the Company at the Real Property.

#### 5.16 ERISA.

(a) Schedule 5.16 contains a true and complete list of all "employee benefit plans," within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and any other bonus, profit sharing, pension, severance, savings, deferred compensation, fringe benefit, insurance, welfare, post-retirement benefit, health, life, stock option, stock purchase, restricted stock, tuition refund, service award, company car, scholarship, relocation, disability, accident, sick, vacation, individual employment, consulting, compensation, incentive, commission, payroll practices, retention, change in control, noncompetition, or other plan, agreement, policy, trust fund, or arrangement (whether written or unwritten, insured or self-insured) established, maintained, sponsored, or contributed to (or with respect to which any obligation has been undertaken) by the Company or any entity that would be deemed a "single employer" with the Company under section 414(b), (c), (m) or (o) of the Internal Revenue Code of 1986, as amended (the "Code") or

section 4001 of ERISA (an "ERISA Affiliate") on behalf of any employee, consultant, director or shareholder of the Company (whether current, former or retired) or their beneficiaries (each a "Plan" and, collectively, the "Plans"). With respect to each Plan, to the extent applicable, true and complete copies of the plan and documents (including amendments) and summary plan description, IRS Form 5500, if any, IRS determination letter and other material documents relating to the Plan have been delivered to TISH. None of the Company, any ERISA Affiliate, or any of their respective predecessors has ever contributed to, contributes to, has ever been required to contribute to, or otherwise participated in or participates in or in any way, directly or indirectly, has or could reasonably be expected to have any liability with respect to any plan subject to Section 412 of the Code, Section 302 of ERISA or Title IV of ERISA, including, without limitation any, "multiemployer plan" (within the meaning of Sections (3) (37) or 4001(a) (3) of ERISA or Section 414(f) of the Code), or any

single employer pension plan.

(b) Each Plan complies in all material respects and has been maintained and operated in all material respects in compliance with its terms and applicable law, including, without limitation, ERISA and the Code. With respect to each of the Plans, except as set forth on schedule 5.16: (i) each Plan intended to qualify under Section 401(a) of the Code is qualified and has received a determination letter from the IRS to the effect that the Plan is qualified under Section 401 of the Code and any trust maintained pursuant thereto is exempt from federal income taxation under Section 501 of the Code and nothing has occurred or is expected to occur through the Closing Date that caused or could cause the loss of such qualification or exemption or the imposition of any penalty or tax liability; (ii) all payments required by any Plan, any collective bargaining agreement or other agreement, or by law with respect to all periods through the closing shall have been made prior to the Closing Date to the extent due and payable on the Closing Date (on a pro rata basis where such payments are otherwise discretionary at year end) or provided for by the Company as applicable, by full accruals as if all targets required by such Plan had been or will be met at maximum levels) on its financial statements; (iii) no claim, lawsuit, arbitration or other action has been threatened, asserted or instituted against the Plans (other than non-material routine claims for benefits, and appeals of such claims), any trustee or fiduciaries thereof, the Company, any ERISA Affiliate, any director, officer, or employee thereof, or any of the assets of any trust of the Plans and the Company is unaware of any facts that could reasonably be expected to give rise to any such claim, lawsuit, arbitration, or other action; (iv) no "prohibited transaction," within the meaning of Section 4975 of the Code and Section 406 of ERISA, has occurred or is expected to occur with respect to the Plan; (v) no Plan is under audit or investigation by the IRS, U.S. Department of Labor, or any other governmental authority and no such completed audit, if any, has resulted in the imposition of any tax or penalty and (vi) with respect to each Plan that is funded mostly or partially through an insurance policy, neither the Company nor any ERISA Affiliate has any liability in the nature of retroactive rate adjustment, loss sharing arrangement or other actual or contingent liability arising wholly or partially out of events occurring on or before the Closing Date.

(c) Except as provided in schedule 5.16(c), the consummation of the transactions contemplated by this agreement will not give rise to any liability including, without limitation, liability for severance pay, unemployment compensation, termination pay, or

withdrawal liability, or accelerate the time of payment or vesting or increase the amount of compensation or benefits due to any employee, director or shareholder of the Company (whether current, former, or retired) or their beneficiaries solely by reason of such transaction or by reason of a termination of employment following such transaction. No amounts payable under any Plan will fail to be deductible for federal income tax purposes by virtue of Sections 280G or 162(m) of the Code.

- (d) Neither the Company nor any ERISA Affiliate maintains, contributes to, or in any way provides for any benefits of any kind whatsoever (other than under Section 4980B of the Code, the Federal Social Security Act, or a plan qualified under Section 401(a) of the Code) to any current or future retiree or terminee. No event, condition, or circumstance exists that could result in an increase of the benefits provided under any Plan based on the Plan documents from the level of benefits for the most recent fiscal year ended before the Closing Date; and none of the Company, any ERISA Affiliate or officer or employee thereof, has made any legally binding promises or commitments to create any additional plan, agreement or arrangement, or to modify or change any existing Plan. Neither the Company nor any ERISA Affiliate has any unfunded liabilities pursuant to any Plan that is not intended to be qualified under Section 401(a) of the Code. No event, condition, or circumstance exists that could reasonably be expected to prevent the amendment or termination of any Plan.
- 5.17 Brokers and Finders. Neither the Stockholders nor the Company has employed any broker or finder or incurred any liability (on its behalf or on behalf of others, including TISH) for any investment banking fees, brokerage fees or commissions or finders fees in connection with the transactions contemplated by this agreement.
- 5.18 Transactions with Affiliates. Except as set forth on schedule 5.18, the Company is not, and since December 31, 1998 has not, engaged, with respect to the Business, in any transaction with any officer or director of the Company, any of their family members, or any entity in which any of them has an interest.
- 5.19 Permits and Licenses. The Company has, and is in compliance, in all material respects, with, all permits, licences, franchises and other authorizations ("Licenses") necessary for the conduct of the Business, and all such Licenses are valid and in full force and effect. All Licenses held by the Company that are material to the Business are listed on schedule 5.19.
- 5.20 Taxes. The Company has timely filed all federal, state and other tax returns required by law to be filed by it, and each of those tax returns is correct and complete in all respects. The Company has paid, or made provision in its financial statements referred to in section 5.7 for payment of, all of its taxes accrued through the date of those financial statements, except that any accrued but unpaid taxes for 1999 and beyond are not reflected in financial statements prepared on a cash basis. There are no claims pending or threatened in writing against the Company for past due taxes. All taxes that are or were required by law to be withheld or collected by the Company have been duly withheld or collected and paid to the proper tax authority. Complete copies of all federal, state and local income tax returns of the

Compar	ıу	for	eac	ch c	of	the	taxable	years	ended		have	e been	delive	ered	to
TISH.	Th	ere	is	no	ac	tion	, suit,	procee	eding,	investigati	on, a	audit,	claim	or	

assessment pending or proposed with respect to any tax liability of the Company; no extension of time within which to file any tax return of the Company has been requested and the return not yet filed; there are no waivers or extensions of any applicable statute of limitations for the assessment or collection of any tax with respect to the Company in effect; there are no tax rulings, requests for rulings, closing agreements or changes of accounting method relating to the Company that could affect its tax liability for any period after the Closing Date; the Company has not filed a consent under section 341(f) of the Internal Revenue Code; the Company will not be required to include in a taxable period ending after the Closing Date taxable income attributable to, but not recognized in, a prior taxable period. Neither the Company nor any predecessor has been included in a consolidated or combined tax return.

#### 5.21 Insurance.

- (a) Schedule 5.21(a) contains a complete list of all insurance policies relating to the Business under which the Company is currently an insured, a named insured or otherwise the principal beneficiary of coverage. True and complete copies of all of those policies have been delivered or made available to TISH. The Company has not received any notice that any of those insurance policies are not in full force and effect. The Company is not in breach or default (including any breach or default with respect to the payment of premiums or the giving of notice), and no event has occurred which, with notice or the lapse of time, would constitute such a breach or default or permit termination or modification, under any those insurance policies. The Company has not received any written notice of cancellation or intent to cancel any of those insurance policies. The Company has paid all premiums due and payable with respect to those insurance policies.
- (b) Schedule 5.21(b) sets forth a list of all claims of the Company relating to the Business that are pending with any insurance company.

#### 5.22 Employees.

- (a) Schedule 5.22 is a true and complete list of all current employees of the Company or any of its affiliates who perform, or who at any time during the twelve months preceding the date of this agreement performed, substantial services with respect to the Business, which list identifies the name and position of each such employee, and the following compensation information for fiscal years 1997, 1998 and 1999: (a) annual base salary, (b) annual bonus, (c) commissions, (d) perquisites, (e) severance, (f) all other items of compensation and (g) date of birth and date of hire.
- (b) Except as set forth on schedule 5.16, the Company does not have any severance policy and no employee of the Business is entitled to any severance payment, either by law or by agreement, upon the termination of his or her employment. No employee of the Business is represented by any union or other collective bargaining agent and there are no collective bargaining or other labor agreements with respect to those employees.

- (c) Except as set forth on schedule 5.22, no employee has threatened in writing, nor to the knowledge of the Stockholders or the Company, intends to terminate his or her employment.
- 5.23 Year 2000 Compliance. Except as set forth on schedule 5.23, the computer systems (including all work stations and other components) used in the Business, both before and after the beginning of the year 2000, (i) are capable of recognizing, processing, managing, representing, interpreting, and manipulating correctly date related data for dates earlier and later than January 1, 2000, including, but not limited to, calculating, comparing, sorting, storing, tagging, and sequencing, without resulting in or causing logical or mathematical errors or inconsistencies in any user-interface functionalities, data storage, data fields, calculations, reports, processing, or any other input or output, (ii) have the ability to provide date recognition for any data element represented without a date, or whose year is represented by only two digits and the ability to automatically function into and beyond the year 2000 without human intervention, (iii) correctly interpret data, dates and time into and beyond the year 2000, (iv) have the ability not to produce non-compliance in existing information, nor otherwise corrupt such data, and (v) have the ability to successfully interface with internal and external applications or systems that have not yet achieved year 2000 compliance during the time in which the systems and such applications and systems co-exist.
- 5.24 Customers. Schedule 5.24 contains a complete list of the twenty largest customers (measured by dollar volume in each case) of the Company during each of the last three fiscal years and during the first month of the current fiscal year, showing, with respect to each, the name and address, dollar volume involved, and the percentage of each of the Company's lines of business which each such customer represented. The Company is not required to provide any bonding or other financial security arrangements in connection with any transactions with any customers in the ordinary course of its business. Except as set forth on schedule 5.24, the Company is not and within the last two years has not been engaged in any material disputes with its customers. To the knowledge of the Stockholders or the Company, except as set forth on schedule 5.24, there has been no loss or threatened loss of any material customer nor are the Stockholders aware of any material customer which is considering termination, nonrenewal or any adverse modification of its arrangements with the Company before the closing or with TISH after the closing.
- 5.25 No Misrepresentation. No representation or warranty by the Stockholders in this agreement (including the schedules and exhibits to this agreement) contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in this agreement (including the schedules and exhibits to this agreement) not misleading.
- 6. Representations and Warranties by TISH. TISH and TOIC jointly and severally represent and warrant to the Stockholders that, as of the date hereof:
- 6.1 TISH's Organization. TISH is a corporation duly organized, validly existing and in good standing under the law of the State of Delaware and

power and authority to enter into and to perform this agreement. TISH is a wholly owned subsidiary of TOIC and was formed for the purpose of consummating the transactions contemplated under this agreement and the Affiliate Acquisition Agreements and has conducted no other business prior to the date hereof. TISH has delivered to the Stockholders a true and correct copy of the certificate of incorporation and bylaws of TISH. The capitalization of TISH immediately after the closing is set forth on schedule 6.1.

- 6.2 Authorization of Agreement. The execution, delivery and performance of this agreement and the Closing Documents by TISH and TOIC have been duly authorized by all requisite corporate action of TISH and TOIC. This agreement constitutes, and the Closing Documents when executed will constitute, the valid and binding obligation of TISH and TOIC enforceable against each of them in accordance with its terms, except as enforceability may be limited by bankruptcy, insolvency or other similar laws affecting the enforcement of creditors' rights in general and subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).
- 6.3 Consents of Third Parties. The execution, delivery and performance of this agreement by TISH or TOIC will not (i) conflict with the certificate of incorporation or by-laws of TISH or TOIC, (ii) conflict with, result in the breach or termination of, or constitute a default under, any lease, agreement, commitment or other instrument, or any order, judgment or decree, to which TISH or TOIC is a party by which it is bound, or (iii) constitute a violation by TISH or TOIC of any law or regulation applicable to it. No consent, approval or authorization of, or designation, declaration or filing with, any governmental authority is required on the part of TISH or TOIC in connection with the execution, delivery and performance of this agreement.
- 6.4 Brokers and Finders. Neither TISH nor TOIC has employed any broker or finder or incurred any liability (on its behalf or on behalf of others, including the Company) for any investment banking fees, brokerage fees or commissions or finders fees in connection with the transactions contemplated by this agreement, except for Veronis, Suhler & Associates, Inc. (the fees and costs of which shall be paid by TISH or TOIC).

#### 6.5 TISI.

- (a) Simultaneously with the closing, TOIC shall transfer to TISH all of the outstanding shares of capital stock of TISI.
- (b) TISI is a corporation duly organized, validly existing and in good standing under the law of the State of Oklahoma and has the full corporate power and authority to carry on its business as it currently is being conducted.

(c) Schedule 6.5(c) contains financial statements consisting of (i) balance sheets of TISI as of December 31, 1997 and December 31, 1998, and the related statements of income and cash flows for the 12 month periods then ended, including all related notes thereto, and (ii) a proforma balance sheet of TISI as of December 31, 1999 and the related

proforma statement of income and cash flow for the 12 month period then ended (after giving effect to pending and completed acquisitions). Except as set forth on schedule 6.5(c), all such financial statements have been prepared in accordance with generally accepted accounting principles applied on a consistent basis and fairly present the financial position of TISI as of the dates indicated and the results of operations of TISI for the periods indicated. All such financial statements have been prepared in accordance with TISI's books and records and show all income and expenses attributable to TISI's business during the respective periods covered by them. Except as set forth on schedule 6.5(c), as of December 31, 1999, TISI had not incurred any material liability other than as disclosed or reserved for on its proforma financial statements as of December 31, 1999 or in connection with contracts and other commitments, obligations and liabilities entered into or incurred in the ordinary course of business. Since December 31, 1999, there has been no material adverse change in the financial condition of TISI.

- (d) Except as set forth on schedule 6.5(d), there is no claim, litigation, proceeding or governmental investigation pending or, to the best knowledge of TISH, threatened, or any order, injunction or decree outstanding against or relating to TISI or TISH which will or is likely to have a material adverse effect on TISI, TISH and their subsidiaries taken as a whole. TISH does not know of any basis for future claims, litigations, proceedings or investigations against TISI or TISH which is likely to have a material adverse effect on TISI, TISH and their subsidiaries taken as a whole. Neither TISI nor TISH is in violation, in any material respect, of any law, regulation, ordinance, or any other requirement of any governmental body or court, including the Fair Credit Reporting Act, arising out of the operation of its business, and no notice has been received by TISI or TISH or any of their respective officers or directors alleging any such violation, in any case where such violation or alleged violation is likely to have a material adverse effect on TISI, TISH and their subsidiaries taken as a whole.
- (e) At the closing, the TISH Shares to be issued will be duly authorized and validly issued, fully paid and nonassessable and free and clear of all Liens, and the percentage ownership of each of the Stockholders shall be as provided on schedule 6.1.
- 6.6 Year 2000 Compliance. The computer systems (including all work stations and other components) used in TISI's business, before and after the beginning of the year 2000, (i) are capable of recognizing, processing, managing, representing, interpreting, and manipulating correctly date related data for dates earlier and later than January 1, 2000, including, but not limited to, calculating, comparing, sorting, storing, tagging, and sequencing,

without resulting in or causing logical or mathematical errors or inconsistencies in any user-interface functionalities, data storage, data fields, calculations, reports, processing, or any other input or output, (ii) have the ability to provide date recognition for any data element represented without a date, or whose year is represented by only two digits and the ability to automatically function into and beyond the year 2000 without human intervention, (iii) correctly interpret data, dates and time into and beyond the year 2000, (iv) have the ability not to produce non-compliance in existing information, nor otherwise corrupt such data, and (v) have the ability to successfully interface with internal and external applications or systems that have not yet achieved year 2000 compliance during the time in which the systems and such applications and systems co-exist.

- 6.7 No Misrepresentation. No representation or warranty by TISH or TOIC in this agreement (including the schedules and exhibits to this agreement) contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in this agreement (including the schedules and exhibits to this agreement) not misleading.
- 7. Further Agreements of the Parties; Nondisclosure and Confidentiality.
- 7.1 Access to Information. Prior to the closing, TISH and its representatives may, upon reasonable request during normal business hours, make such in vestigation of the Real Property, assets and operations of the Business as it may desire, and the Stockholders shall give to TISH and to its counsel, accountants and other representatives, upon reasonable notice, full access during normal business hours throughout the period prior to the closing to all of the Real Property, assets, books, commitments, agreements, records and files of the Company relating to the Business and the Real Property and the Stockholders shall furnish to TISH during that period all documents and copies of documents and information concerning the Business and the Real Property as TISH reasonably may request. TISH shall hold, and shall cause its representatives to hold, all such information and documents and all other information and documents delivered pursuant to this agreement confidential and, if the transactions contemplated by this agreement are not consummated for any reason, shall return to the Stockholders all such information and documents and any copies as soon as practicable, and shall not disclose or use any such information (that has not previously been disclosed by a party other than TISH) to any third party unless required to do so pursuant to a request or order under applicable laws and regulations or pursuant to a subpoena or other legal process. TISH's obliga tions under this section shall survive the termination of this agreement.
- 7.2 Conduct of the Business Pending the Closing. Until the closing, the Stockholders shall cause the Company to operate the Business in the ordinary course in a manner consistent with past practices and:
  - (a) the Stockholders shall promptly notify TISH in writing

of, and furnish any information that TISH reasonably may request with respect to, (i) any claim, litigation, proceeding or governmental investigation threatened in writing by or against the Company relating to the Business or any material development with respect to any such claim, litigation, proceeding or governmental investigation, (ii) the occurrence of any event or the existence of any state of facts that would result in any of the Stockholders' representations and warranties not being true as of the Closing Date, and (iii) any other occurrence of any kind materially adversely affecting the Business or the Assets;

(b) the Company shall not (i) grant or agree to grant any general increase in the rates of salaries or compensation of its employees, or any specific increase to any such employee whose total salary or compensation after the increase would be at an annual rate in excess of \$30,000, or any increase or amend any benefits of employees or agents of the Business (including, without limitation, increase in the pension, retirement or other employment benefits of the employees of the Business), (ii) amend, adopt or terminate any Plans or (iii) enter

into any employment, bonus, deferred compensation or any other agreement with any employee or agent;

- (c) the Company shall use reasonable efforts, consistent with its past practices, (i) to preserve the business organization of the Business intact and to preserve the goodwill and business of those having business relations with the Business, (ii) to retain the services of the employees of the Business, and (iii) to preserve all trademarks, trade names, logos and copyrights and related registrations of the Business;
- (d) the Company shall not (i) enter into or renew any agreement, commitment or lease which, if entered into prior to the date of this agreement, would have been required to be included in schedule 5.10, (ii) cause or take any action to allow any lease, agreement or commitment relating to the Business to lapse (other than in accordance with its terms), to be modified in any material adverse respect, or otherwise to become impaired in any manner, (iii) amend, terminate, cancel or compromise any material claim relating to the Business, or waive or allow to lapse or terminate any other rights of substantial value to the Business or (iv) make or commit to make any capital expenditures which, individually or in the aggregate, involve payments in excess of \$50,000;
- (e) except in the ordinary course and substantially consistent with past practice, the Company shall not (i) enter into any transaction or voluntarily incur any liability or obligation that is material to the Business or (ii) sell or transfer any of the assets used in the Business, other than assets that have worn out or been replaced with other assets of equal or greater value or assets that are no longer needed in the operation of the Business;

- (f) the Company shall not make any material change in the manner in which the Business is operated or the accounting principles or practices employed in connection with the Business;
- (g) the Company shall duly comply, in all material respects, with all laws, ordinances, orders, injunctions and decrees applicable to the operation of the Business;
- (h) the Company shall maintain all of the tangible assets used in the Business in customary repair, maintenance and condition, except to the extent of normal wear and tear, and the Company shall replace any items of equipment at time intervals consistent with past practices;
- (i) the Company shall maintain insurance on the tangible assets used in the Business and on the Business in accordance with past practices;
- (j) the Company shall not take any action to accelerate collection of accounts receivable or to defer payment of liabilities, except in the ordinary course consistent with past practices; and
- (k) the Company may make cash distributions to the Stockholders provided such distributions do not result, as of the closing, in current liabilities exceeding current assets.
- 7.3 Other Action. No party to this agreement shall take any action that would result in any of its representations and warranties not being true as of the Closing Date. Each of the parties to this agreement shall use its best efforts to cause the fulfillment at the earliest practicable date of all of the conditions to the obligations of the parties to consummate the transactions under this agreement. The Stockholders shall cause the Company to fulfill any obligations of the Company under this agreement.
- 7.4 Consents. The Stockholders shall use their best efforts to obtain at the earliest practicable date, in form and substance reasonably satisfactory to TISH, all consents and approvals referred to on schedule 5.3 without any condition adverse to TISH or the Company or the operation of the Business after the Closing Date.
- 7.5 Expenses. TISH and the Stockholders shall bear their own respective expenses incurred in connection with this agreement and in connection with all obligations required to be performed by each of them under this agreement.
- 7.6 Transfer Taxes. The Stockholders shall pay any transfer taxes payable in connection with the transfer of the Shares pursuant to this agreement.
  - 7.7 Covenants Against Competition, Solicitation and Disclosure.

- (a) To accord to TISH the full value of its acquisition, during the Noncompete Term, as defined below, the Stockholders shall not, directly or indirectly, engage, or be interested in (as owner, stockholder, partner, member, manager, lender, agent or otherwise) or provide any services to any business or entity that engages, anywhere in the United States, in the operation of pre-employment investigation and screening services, except pursuant to any employment, consulting or transition services agreements entered into between the Stockholders and TISH.
- (b) During the Noncompete Term, the Stockholders shall not, directly or indirectly, employ or solicit for employment or consulting, on its own behalf or on behalf of any other person or entity, or otherwise encourage the resignation of, any employee of the Company who provided substantial services to the Business at any time during the one year period ending on the Closing Date.
- (c) The Stockholders shall not at any time hereafter disclose to anyone, or use in competition with the Business, any information relating to the Business, which is known or becomes known to the Stockholders, including, without limitation, trade secrets, know-how, financial statements and data, contracts (whether oral or written), customer and advertiser lists, rate schedules, pricing policies, marketing plans and strategies, and business acquisition plans ("Confidential Information") [and shall not provide to anyone in competition

with the Business any information derived from the Stockholders' operation of the Excluded Business]. Confidential Information shall not include information which the Stockholders can demonstrate has become generally available to the public other than as a result of a disclosure by the Stockholders.

- (d) The Stockholders acknowledge that the remedy at law for breach of the provisions of this section 7.7 will be inadequate and that, in addition to any other remedy TISH may have, it shall be entitled to an injunction restraining any breach or threatened breach, without any bond or other security being required and without the necessity of showing actual damages. If any court construes any covenant in this section 7.7 to be unenforceable in any respect, the court may reduce the duration or area to the extent necessary so that the provision is enforceable, and the provision, as reduced, shall then be enforceable.
- (e) "Noncompete Term" shall mean the later of the three year period after the Closing Date or, with respect to each Stockholder, the three year period after the termination of such Stockholder's employment with TISH pursuant to an employment agreement referred to in section 7.9.
- 7.8 Stockholders' Agreement. At the closing, TISH, TOIC and the Stockholders shall execute and deliver a stockholders' agreement in the form of exhibit VII.

- 7.9 Employment Agreements. At the closing, TISH and the Stockholders shall execute and deliver employment agreements in the form of exhibit VIII.
- 7.10 Escrow Agreement. At the closing, TISH and the Stockholders shall execute and deliver the Escrow Agreement.
- 7.11 Transfer of TISI. Simultaneously or prior to the closing, TOIC shall transfer all of the outstanding shares of TISI to TISH and ensure the completion of all the actions and transactions described in schedule 7.11.
- [7.12 License. The Stockholders hereby agree to provide to TISH periodically (but in no event less frequently than monthly) all incident or theft information arising in connection with the Retained Assets (the "Licensed Data") and hereby grant to TISH a perpetual, non-exclusive, royalty free, worldwide license to use the Licensed Data in connection with the Business, but only to the extent that the Stockholders have the right to so license all or a portion of the Licensed Data. During the Noncompete Term, the Stockholders agree not to use, or license to any other party to use, the Licensed Data to a third party in connection with any business competitive with the Business. The Stockholders will use commercially reasonable efforts to obtain the consent of the owner of any Licensed Data (to the extent the Stockholders are not the owner) to allow such data to be licensed to TISH pursuant hereto.]
- 7.13 Interim Financial Statements. The Stockholders shall promptly (but, in any event, not later than the date such statements customarily are completed by the Company) deliver to TISH copies of any monthly or quarterly financial statements or other

reports relating to the Business that may be prepared by the Company during the period from the date of this agreement to the Closing Date. All such financial statements shall be prepared from the books and records of the Business and shall show all income and expenses attributable to the Business and shall fairly present the financial position and results of operations of the Business as of and for the periods indicated. The Stockholders shall also furnish to TISH any other information concerning the financial and operating condition of the Business as TISH from time to time reasonably may request.

- 7.14 Further Assurances. At any time and from time to time after the Closing Date each party shall, without further consideration, execute and deliver to the other such other instruments of transfer and assumption and shall take such other action as the other may reasonably request to carry out the transfer of the operations, rights and assets and assumption of liabilities contemplated by this agreement.
- 7.15 Post-Closing Services. For a period of [six] months after the closing, each of the Stockholders who is not otherwise entering into employment or consulting agreements with TISH at the closing agrees to continue to work for

the Business on a basis and in a manner consistent with past practices with respect to that Stockholder, in order to transfer the Business to TISH (without any obligation to incur expenses), subject to the supervision and instructions of the officers of TISH, and such Stockholders shall receive the same salary and benefits as they had received immediately prior to the date of this agreement [as described on schedule 7.15]. During such period, the Stockholders who are entering into employment or consulting agreements with TISH at the closing agree to comply with the provisions of such employment and consulting agreements. At TISH's option, the Stockholders' services under this agreement may be terminated at any time on 5 business days' prior notice, subject, however, to the terms of any employment or consulting agreements.

7.16 Payment of Taxes. The Stockholders shall pay all taxes of the Company relating to periods through the Closing Date.

[Insert Stores Protective Association, Inc. Riders]

- 8. Conditions to Closing.
- 8.1 Conditions Precedent to Obligations of TISH and TOIC. TISH's and TOIC's obligation to consummate the transactions under this agreement is subject to the fulfillment, prior to or at the closing, of each of the following conditions (any or all of which may be waived by TISH and TOIC):
- (a) all representations and warranties of the Stockholders to TISH and TOIC shall be true and correct in every material respect as of the time of the closing, with the same effect as though those representations and warranties had been made again at and as of that time;
- (b) the Stockholders shall have performed and complied with all obligations and covenants required by this agreement to be performed or complied with by them prior to or at the closing;
- (c) TISH and TOIC shall have been furnished with a certificate (dated the Closing Date and in form and substance reasonably satisfactory to TISH) executed by the Stockholders, certifying to the fulfillment of the conditions specified in sections 8.1(a) and 8.1(b);
- (d) the transactions contemplated by each Affiliate Acquisition Agreement shall be consummated at the closing;
- (e) TISH and TOIC shall have been furnished with an opinion of Jenner & Block, counsel to the Stockholders in the form of exhibit IX;
- (f) the Stockholders shall have duly received, without any condition adverse to the Business, TISH or TOIC, all consents and approvals listed on schedule 5.3, each in form and substance reasonably satisfactory to TISH;

- (g) there shall not be in effect any injunction or restraining order issued by a court of competent jurisdiction in an action or proceeding against the consummation of the transactions pursuant to this agreement; and
- (h) the Stockholders or the Company, as the case may be, shall have duly executed and delivered to TISH the agreements referred to in sections 2.1(a) (Escrow Agreement), 7.8 (stockholders' agreement) and 7.9 (employment agreements), and such agreements shall be in full force and effect.
- (i) Merchants Security Exchange ("MSE") shall have duly authorized the sale by United States Mutual Association, Inc. contemplated by the Affiliate Acquisition Agreement relating thereto;
- (j) MSE and TISH shall have entered into a services agreement on terms acceptable to TISH; and.
- (k) the Company shall have made the Pre-Closing Distribution to the Stockholders.
- 8.2 Conditions Precedent to Obligations of the Stockholders. The Stockholders obligations to consummate the transactions under this agreement are subject to the fulfillment, prior to or at the closing, of each of the following conditions (any or all of which may be waived by the Stockholders):
- (a) all representations and warranties of TISH and TOIC shall be true and correct in every material respect as of the time of the closing, with the same effect as though those representations and warranties had been made again at and as of that time;
- (b) TISH and TOIC shall have performed and complied with all obligations and covenants required by this agreement to be performed or complied with by them prior to or at the closing;
- (c) the Stockholders shall have been furnished with a certificate (dated the Closing Date and in form and substance reasonably satisfactory to the Stockholders) executed by an officer of TISH, certifying to the fulfillment of the conditions specified in sections 8.2(a) and 8.2(b);
- (d) the transactions contemplated by each Affiliate Acquisition Agreement shall be consummated at the closing;
- (e) the Stockholders shall have been furnished with an opinion of Brian A. Meyer, Esq., general counsel to TISH, in the form of exhibit X:
- (f) there shall not be in effect any injunction or restraining order issued by a court of competent jurisdiction in an action or proceeding against the consummation of the transactions pursuant to this

- (g) TISH or TISI shall have duly executed and delivered to the Stockholders or the Company, as the case may be, the agreements referred to in section 8.1(h);
- (h) the conditions set forth in sections 8.1(i) and 8.1(j) shall be fulfilled; and
- (i) the other transactions set forth on schedule 7.11 have been completed.
  - 9. Documents to Be Delivered at Closing.
- 9.1 Documents to Be Delivered by the Stockholders. At the closing, the Stockholders shall deliver to TISH the following:
- (a) stock certificates representing the Shares, duly endorsed in blank or accompanied by stock transfer powers, [with all requisite stock transfer tax stamps attached];
- (b) the written resignations of the directors and officers of the Company listed on schedule 9.1(b);
  - (c) the certificate referred to in section 8.1(c);
  - (d) the opinion referred to in section 8.1(e);
- (e) the agreements referred to in sections 2.1(a), 7.8 and 7.9;
- (f) the certifying statement issued by the Company pursuant to Treasury Regulations section  $1.897-2\,(h)\,(1)$ , dated the Closing Date, in form reasonably satisfactory to TISH and its counsel.
- 9.2 Documents To Be Delivered by TISH. At the closing, TISH shall deliver to the Stockholders the following:
  - (a) the wire transfer of funds referred to in section 2.1;
- (b) certificates representing the number of shares of Class B Non-Voting Common Stock and Class C Non-Voting Common Stock of TISH referred to in section 2.1.
- (c) a copy of resolutions of the board of directors of TISH and TOIC authorizing the execution, delivery and performance of this agreement by TISH and TOIC, and a certificate of its secretary or assistant secretary, dated the Closing Date, that such resolutions were duly adopted and are in full force and effect;

- (d) the certificate referred to in section 8.2(c);
- (e) the opinion referred to in section 8.2(e); and
- (f) the agreements referred to in section 2.1(a), 7.8 and 7.9.
- 9.3 Notice to Internal Revenue Service . Immediately prior to the closing, the Stockholders shall cause the Company to provide to the Internal Revenue Service the notice required by Treasury Regulations section 1.897-2(h)(2) (which shall include any supplemental statements required by Treasury Regulations section 1.897-2(h)(5)) with regard to the certifying statement referred to in section 9.1(f).
  - 10. Survival of Representations and Warranties; Indemnification.

#### 10.1 Survival.

(a) All representations, warranties and agreements by the Stockholders shall survive the closing (but only for the respective periods set forth in section 10.3 below) notwithstanding any investigation at any time by or on behalf of TISH or TOIC, and shall not be considered waived by TISH's or TOIC's consummation of the acquisition

contemplated by this agreement with knowledge of any breach or misrepresentation by the Stockholders.

- (b) All representations, warranties and agreements by TISH and TOIC shall survive the closing (but only for the respective periods set forth in section 10.3 below) notwithstanding any investigation at any time by or on behalf of the Stockholders, and shall not be considered waived by the Stockholders' consummation of the acquisition contemplated by this agreement with knowledge of any breach or misrepresentation by TISH or TOIC.
  - 10.2 Indemnification; Exclusive Remedy.
- (a) Subject to the other sections of this Article 10, the Stockholders shall, jointly and severally, indemnify and hold harmless TISH and TOIC against all loss, liability, damage or expense (including reasonable fees and expenses of counsel, whether involving a third party or between the parties to this agreement) TISH or TOIC may suffer, sustain or become subject to as a result of (i) any breach of any warranty of the Stockholders contained in this agreement, or any misrepresentation by the Stockholders (determined without regard to any materiality, material adverse effect or knowledge qualifiers), or any claim by a third party which, without regard to the merits of the claim, would constitute such a breach or misrepresentation, (ii) any breach of any covenant or other agreement of the Stockholders contained in this agreement or any claim by a third party which, without regard to the merits of the claim,

would constitute such a breach, (iii) any obligations, liabilities, agreements or commitments (whether accrued, absolute, contingent or otherwise) relating to or arising out of the operation of the Business prior to the Closing Date (including, but not limited to, any liability with respect to any claim, litigation or proceeding arising out of such operation of the Business prior to the Closing Date) that are not expressly Permitted Liabilities, (iv) any claims made by MSE or any person claiming to have an interest in the capital stock or assets of the Company arising out of the transactions contemplated by this agreement (but not any claims relating to any business or other relationship between MSE and TISH or TOIC or any of their affiliates, which are unrelated to the transactions contemplated by this agreement), or (v) any tax or other cost or expense incurred by the Company in connection with making the Pre-Closing Distribution [and any claims or liabilities with respect to the Excluded Business.]

(b) Subject to the other sections of this Article 10, TISH and TOIC shall, joint and severally, indemnify and hold harmless the Stockholders against all loss, liability, damage or expense (including reasonable fees and expenses of counsel, whether involving a third party or between the parties to this agreement) the Stockholders may suffer, sustain or become subject to as a result of (i) any breach of any warranty contained in this agreement or any misrepresentation by TISH or TOIC, or any claim by a third party which, without regard to the merits of the claim, would constitute such a breach or misrepresentation or (ii) any breach of covenant or other agreement contained in this agreement by TISH or TOIC, or any claim by a third party which, without regard to the merits of the claim, would constitute such a breach.

(c) [Except for the ability to seek injunctive or similar equitable relief to enforce affirmative covenants], the remedy of TISH, TOIC and their affiliates against the Stockholders pursuant to this agreement, any documents or instruments pursuant to this agreement, or at law, equity or otherwise, is limited solely to the indemnification provided in this section 10 and is the only remedy of any of them against the Stockholders, and TISH and TOIC, for themselves and their affiliates, hereby expressly waive all rights and remedies whatsoever, whether by statute, rule, regulation, in tort or otherwise, against the Stockholders except for the right of indemnification under this section 10.2, provided that the foregoing shall not apply to (x) claims of fraud, (y) the covenants set forth in sections 7.7, 7.12, 7.14 and 7.15 and (z) the employment and consulting agreements. TISH and TOIC, for themselves and their affiliates, acknowledge that (i) the Stockholders have not made any representations or warranties other than as expressly set forth in section 5 hereof, (ii) TISH, TOIC and their affiliates have made their own independent examination, investigation, analysis and evaluation of the Company and the Stockholders and have undertaken such due diligence as they have deemed adequate, which examination, investigation, analysis, evaluation and due diligence shall not limit or diminish in any manner the representations and warranties and obligations of the Stockholders under this agreement, and (iii) damages of TISH, TOIC and their affiliates, if any, may be greater than the

funds in escrow pursuant to the Escrow Agreement.

- (d) The Stockholders acknowledge that (i) neither TISH nor TOIC have made any representations or warranties other than as expressly set forth in section 6 hereof, and (ii) the Stockholders have made their own independent examination, investigation, analysis and evaluation of TISH and TOIC and have undertaken such due diligence as they have deemed adequate, which examination, investigation, analysis, evaluation and due diligence shall not limit or diminish in any manner the representations and warranties and obligations of TISH and TOIC under this agreement.
- 10.3 Notices of Claims. No party shall be liable for misrepresentation or breach of warranty except to the extent that notice of a claim is given by another party in writing on or prior to April 30, 2001, except that, (a) for misrepresentation or breach of warranty under sections 5.2, 5.5 or 5.17, notice of a claim must be given in writing to the Stockholders within ten (10) years after the Closing Date, (b) with respect to misrepresentation or breach of warranty relating to taxes and ERISA, notice of a claim must be given in writing to the Stockholders no later than three months after the expiration of the statute of limitation applicable to the tax involved, (c) with respect to misrepresentation or breach of warranty under section 5.15, notice of a claim must be given in writing to the Stockholders within six years after the Closing Date, and (d) for misrepresentation or breach of warranty under sections 6.2, 6.4 and 6.5(e) notice of a claim must be given in writing to TISH or TOIC within ten (10) years after the Closing Date. Prompt written notice upon assertion of a claim that may be indemnified must be given to the party liable for misrepresentation or breach of warranty, but the failure to provide such notice shall not relieve or otherwise affect the obligation of the party liable to provide indemnification hereunder, except to the extent that any Losses (as hereinafter defined) directly resulted from or were caused by such failure.
- 10.4 Basket. Neither the Stockholders nor TISH shall be liable for misrepresentation or breach of warranty under this agreement unless and until the aggregate amount of losses, liabilities, damages and expenses incurred by an indemnified party ("Losses") as a result of all misrepresentations and breaches of warranty under this agreement, together with all such Losses incurred by such party or its affiliates under the Affiliate Acquisition Agreements, exceeds \$75,000 in the aggregate (in which event it shall be liable for the full amount of all Losses). No claim shall be asserted by a party against another party unless such claim, when aggregated with other claims asserted under the Affiliate Acquisition Agreements at the same time, involve at least \$10,000 of claimed Losses. This provision shall not apply to any misrepresentation or breach of warranty under sections 5.5, 5.20 and 6.5(e), or to claims asserted under sections 10.2(a)(ii), (iii), (iv) or (v) or 10.2(b)(ii). In calculating the amount of the loss, liability, damage or exposure to TISH or the Stockholders for misrepresentation or breach of warranty by the other which is indemnified under section 10.2, all materiality, material adverse effect and knowledge qualifiers set forth in a representation or warranty that has been misrepresented or breached shall be disregarded.

10.5 Cap on Liability. Notwithstanding anything to the contrary in this agreement, the aggregate liability of the Stockholders and the Affiliates to TISH for misrepresentation or breach of warranty under this agreement and the Affiliate Acquisition Agreements shall be limited to \$3,000,000 ("Cap"); provided that such Cap shall not apply to breach of sections 5.2, 5.5, 5.20 nor to any claim asserted under section 10.2(a)(ii), (iii), (iv) or (v) or to breach of, or any claim asserted under, the comparable provisions of the Affiliate Acquisition Agreements. All claims shall be first asserted against any amounts held in escrow pursuant to the Escrow Agreement (the "Escrow"), and any claims to which the Cap applies may only be asserted against funds held in the Escrow, there being no recourse against the Stockholders directly for such claims, except that in the event that any amounts held in the Escrow are reduced or reserved against to pay Non-Cap Claims (as defined below), the Stockholders shall be jointly and severally liable for claims to which the Cap applies up to the amount that was paid or reserved against in connection with the Non-Cap Claims. "Non-Cap Claims" shall mean claims for indemnification under this agreement and the Affiliate Acquisition Agreements that are not subject to the Cap.

#### 10.6 Third Party Claims.

(a) If any party entitled to be indemnified pursuant to section 10.2 (an "Indemnified Party") receives notice of the assertion by any third party or any claim or of the commencement by any such third person of any action (any such claim or action, an "Indemnifiable Claim") with respect to which another party hereto (an "Indemnifying Party") is or may be obligated to provide indemnification, the Indemnified Party shall promptly notify the Indemnifying Party in writing (the "Claim Notice") of the Indemnifiable Claim; provided, that the failure to provide such notice shall not relieve or otherwise affect the obligation of the Indemnifying Party to provide indemnification hereunder, except to the extent that any Losses directly resulted from or were caused by such failure.

(b) The Indemnifying Party shall have thirty (30) days after receipt of the Claim Notice to undertake, conduct and control, through counsel of its own choosing, and at its expense, the settlement or defense thereof, and the Indemnified Party shall cooperate with the Indemnifying Party in connection therewith; provided that (i) the Indemnifying Party shall permit the Indemnified Party to participate in such settlement or defense through counsel chosen by the Indemnified Party (subject to the consent of the Indemnifying Party, which consent shall not be unreasonably withheld), provided that the fees and expenses of such counsel shall not be borne by the Indemnifying Party, and (ii) the Indemnifying Party shall not settle any Indemnifiable Claim without the Indemnified Party's consent (which consent shall not be unreasonably withheld). So long as the Indemnifying Party is vigorously contesting any such Indemnifiable Claim in good faith, the Indemnified Party shall not pay or settle such claim without the Indemnifying Party's consent, which consent shall not be unreasonably withheld.

(c) If the Indemnifying Party does not notify the Indemnified Party within thirty (30) days after receipt of the Claim Notice that it elects to undertake the defense of the Indemnifiable Claim described therein, the Indemnified Party shall have the right to contest, settle or compromise the Indemnifiable Claim in the exercise of its reasonable discretion; provided, that the Indemnified Party shall notify the Indemnifying Party of any compromise or settlement of any such Indemnifiable Claim.

#### 11. Miscellaneous.

- 11.1 Entire Agreement. This agreement (including the exhibits and schedules hereto, which shall be deemed an integral part of this agreement) contains, and is intended as, a complete statement of all of the terms of the arrangements between the parties with respect to the matters provided for, supersedes any previous agreements and understandings between the parties with respect to those matters (including, but not limited to, the letter of intent dated August 9, 1999), and cannot be changed or terminated orally.
- 11.2 Governing Law. This agreement shall be governed by and construed in accordance with the law of the State of New York applicable to agreements made and to be performed entirely in New York.
- 11.3 Headings. The section headings of this agreement are for reference purposes only and are to be given no effect in the construction or interpretation of this agreement.
- 11.4 Notices. All notices and other communications under this agreement shall be in writing and shall be deemed given when delivered personally or mailed by registered mail, return receipt requested or faxed to the parties at the following addresses (or to such other address as a party may have specified by notice given to the other party pursuant to this provision):

(a)	If to	the	Stockholders,	addressed	to	them	at:
	Atten	tion	 :				
	Fax N	umbei	 r:				
	with a	a coj	py to:				

Attention:
Fax Number:

\_\_\_\_\_\_

(b) If to TISH or TOIC, addressed to it at:
 The Official Information Company
 250 West 57th Street,
 Suite 2421
 New York, New York 10019
 Attention:
 Fax Number: (212) 247-0026

with a copy to:
Steven L. Kirshenbaum, Esq.
Proskauer Rose LLP
1585 Broadway
New York, New York 10036
Fax Number: (212) 969-2900

- 11.5 Separability. If any provision of this agreement is held to be invalid or unenforceable, the balance of this agreement shall remain in effect.
- 11.6 Waiver. Any party may waive compliance by another with any of the provisions of this agreement. No waiver of any provision shall be construed as a waiver of any other provision. Any waiver must be in writing and must be signed by the party waiving the provision.
- 11.7 Assignment. TISH may assign its rights under this agreement simultaneously with or after the closing to any subsidiary or affiliate of TISH that has assumed (in a writing reasonably satisfactory in form and substance to the Stockholders) TISH's obligations under this agreement, but no such assignment shall relieve TISH of its obligations under this agreement. In addition, TISH, or any subsidiary or affiliate of TISH to whom this agreement has been assigned in whole or in part, may assign its rights under this agreement to its

lenders. This agreement will be binding upon, inure to the benefit of and be enforceable by the parties and their respective successors and permitted assigns.

- 11.8 Arbitration.
  - (a) Any dispute or controversy arising under or in connection

with this agreement other than under the provisions of section 2.3(b), shall be settled exclusively by arbitration to be held in the City of Chicago in accordance with the rules of the American Arbitration Association then in effect; provided that any claim of breach with regard to section 7.7 may, at the option of TISH, be enforced in court. The arbitral tribunal shall be composed of three arbitrators, one of which shall be appointed by TISH, one of which shall be appointed by the Stockholders and one of which shall be appointed by the arbitrators appointed by TISH and the Stockholders. As part of their award, the arbitrators shall make a fair allocation between the parties of the fee and expenses of the American Arbitration Association and the cost of any transcript, taking into account the merits of their claims and defenses. Either party shall be entitled to the pre-trial discovery it would have been entitled to if the proceeding had been brought in a New York court under the Civil Practice Law and Rules. Judgment may be entered on the arbitrators' award in any court having jurisdiction, and the parties irrevocably consent to the jurisdiction of the Chicago courts for that purpose. The parties waive personal service in connection with any such arbitration; any process or other papers under this provision may be served by registered mail, return receipt requested, or by personal service, provided a reasonable time for appearance or response is allowed. The arbitrators may grant injunctive or other relief.

(b) The courts of the State of Illinois in Cook County and the United States District Court for the Northern District of Illinois shall have jurisdiction over the parties with respect to any dispute or controversy arising under or in connection with section 7.7 and, by execution and delivery of this agreement, each of the parties to this agreement submits to the jurisdiction of those courts, including but not limited to, the in personam and subject matter jurisdiction of those courts, waives any objection to such jurisdiction on the grounds of venue or forum non conveniens, the absence of in personam or subject matter jurisdiction and any similar grounds, consents to service of process by mail (in accordance with section 11.4) or any other manner permitted by law, and irrevocably agrees to be bound by any judgment rendered thereby in connection with section 7.7. These consents to jurisdiction shall not be deemed to confer rights on any person other than the parties to this agreement.

- 11.9 No Third Party Beneficiaries. This agreement does not create, and shall not be construed as creating, any rights in favor of any person not a party to this agreement.
- 11.10 Publicity. Neither TISH nor the Stockholders, nor their respective affiliates, shall issue or cause the publication of any press release or other public announcement with respect to the transactions contemplated by this agreement without the consent of the other party, which consent shall not be unreasonably withheld or delayed, except as may be required by law or the regulations or policies of any securities exchange, in which case the party required to make the release or statement shall allow the other party reasonable time to comment on such release or statement in advance of such issuance.

		TOIC hereby guarantees the full and
timely perform	mance of all covenants and	obligations of TISH hereunder.
counterparts,		agreement may be executed in titute the same instrument.
		TISI HOLIDINGS, INC.
		By:
		Name: Title:
		THE OFFICIAL INFORMATION COMPANY
		By:
		Name: Title:
		[Stockholder]

# ASSET ACQUISITION AGREEMENT

March , 2000

The parties to this agreement are \_\_\_\_\_\_, a \_\_\_\_\_ corporation ("Transferor"), the parties listed on exhibit I (collectively, the "Stockholders"), TISI Holdings, Inc., a Delaware corporation ("TISH"), which is a wholly-owned subsidiary of TOIC (hereafter defined) and The Official Information Company, a Delaware corporation ("TOIC").

Transferor operates the business of pre-employment investigation and screening (the "Business"). [Transferor also operates the business of civil recovery and refund checking (the "Excluded Business")]. The parties have agreed upon the transfer to TISH of substantially all of Transferor's business and assets relating to the Business, on the terms set forth in this agreement. The affiliates of Transferor listed on exhibit II (the "Affiliates") have entered into acquisition agreements with TISH on the date hereof (the "Affiliate Acquisition Agreements") whereby either the assets or stock of each of the Affiliates will be transferred to TISH on the terms set forth in the respective Affiliate Acquisition Agreements.

TOIC owns all of the outstanding shares of capital stock of Total Information Services, Inc., an Oklahoma corporation ("TISI"), which, among other things, conducts a business similar to the Business. In connection with the transactions contemplated by this agreement, TOIC is transferring and contributing all of such shares of TISI to TISH.

Accordingly, it is agreed as follows:

- 1. Transfer and Acquisition of Assets.
- 1.1 Assets to be Transferred and Acquired. At the closing referred to in section 3, Transferor shall assign and transfer to TISH, and TISH shall acquire from Transferor, all of Transferor's business and assets as they exist on the Closing Date referred to in section 3, but excluding solely the Excluded Assets referred to in section 1.2 (the assets to be sold and assigned to TISH being referred to collectively as the "Assets"). The Assets include, but are not limited to, the following (subject to section 1.2):
- (a) the databases (and all of Transferor's rights to the information contained therein) used in connection with the Business and all computer software and all trademarks, trade names, patents, trade secrets and logos (including registrations and applications for registration of any of them) now or previously used by Transferor in connection with the Business, including those listed on schedule 4.11, together with the good will of the business associated with those trademarks, trade names, patents, trade secrets and logos;

all of Transferor's rights in copyrights that relate to the Business (including

registrations and

applications for registration of any copyrights); and all other intangible property and proprietary rights that relate to the Business;

- (b) all of Transferor's rights under agreements, commitments, orders and leases, to the extent that they relate to the Business and remain unperformed or unfulfilled on, or by their terms continue after, the Closing Date, including, but not limited to all of Transferor's rights under (i) the agreements, commitments, orders and leases listed on schedule 4.8 (except as otherwise provided on that schedule), (ii) all other agreements, commitments and orders that were entered into prior to the date of this agreement in the ordinary course of business and were not required to be referred to on schedule 4.8 and (iii) all other agreements, commitments and orders that are entered into after the date of this agreement in the ordinary course of business and consistent with section 6.2;
- (c) all of Transferor's office supplies, stationery, forms, labels, and similar supplies that relate to, or are used in, the Business;
- (d) all files and other records and data relating to the operations of the Business;
- (e) all equipment (including computers and office equipment), vehicles, furniture, fixtures, supplies, leasehold improvements and improvements on land, and all other tangible personal property and fixed assets, wherever located, that are owned by Transferor and used in or relate to the Business;
- (f) all of Transferor's accounts receivable arising out of the operation of the Business;
- (g) all of Transferor's prepaid expenses, deposits and other current assets relating to the Business;
- (h) all of Transferor's claims against third parties arising out of the operation of the Business;
- (i) all of Transferor's rights to the post office boxes, telephone numbers and facsimile numbers used in the Business;
- (j) all real property, easements, rights-of-way and leasehold and other interests in real property, buildings and improvements owned, leased or used by Transferor in connection with the Business, including, but not limited to, to properties listed on schedule 4.4(c); and
- (k) all of Transferor's rights with respect to Internet domain names and web sites and the content of web sites that relate to the

- 1.2 Excluded Assets. The following assets shall be retained by Transferor and shall not be assigned or transferred to TISH (the "Excluded Assets"):
- (a) all cash, cash investments, certificates of deposit, commercial paper, treasury bills and notes, money market accounts and other marketable securities except to the extent required to maintain current assets at a level equal to current liabilities, as contemplated under section 2.3;
- (b) Transferor's rights under any agreement, commitment or order as to which consent to assignment is required but has not been obtained or which is indicated on schedule 4.8 as not being assigned to TISH, subject to any benefit given to TISH under section 6.4;
- (c) the assets referred to on schedule 1.2(c), [which schedule will refer to income tax returns and financial statements, minute books and stock records; assets relating to Transferor's other lines of business [including the Excluded Business], if any];
- (d) all loans or other receivables due from any Stockholder or affiliate thereof; and
- (e) all the shares of capital stock held by Transferor in United States Mutual Association, Inc.
  - 2. Purchase Price.
    - 2.1 Amount of Consideration. As full consideration for the Assets:
- (a) at the closing, TISH shall pay to Transferor, by wire transfer of immediately available funds, the sum of \$\_\_\_\_\_, subject to adjustment as provided in section 2.3, which shall be payable as follows: (i) \$\_\_\_\_\_ to Transferor, (ii) \$\_\_\_\_\_ into a trust account established by Transferor and the Affiliates pursuant to an agreement of even date herewith, as more fully described on exhibit III, and (iii) \$\_\_\_\_\_ to the escrow agent under an escrow agreement in the form of exhibit IV ("Escrow Agreement");
- (b) at the closing, TISH shall cause to be issued to Transferor \_\_\_\_ shares of Class B Non-Voting Common Stock of TISH and \_\_\_\_ shares of Class C Non-Voting Common Stock of TISH as more fully described on exhibit V (collectively, the "TISH Shares");
- (c) at the closing, TISH shall assume, and agree to pay, perform and discharge solely (the "Assumed Liabilities"):
  - (i) all accounts payable and accrued expenses incurred

in the ordinary course of business, solely if and to the extent reflected or reserved for on the Closing Balance Sheet;

(ii) all obligations under the agreements, commitments and orders relating to the Business that are assigned to TISH pursuant to section 1.1, in each case to the extent that they relate to the period after the Closing Date or are liabilities fully reserved on the Closing Balance Sheet;

(iii) any Unknown Report Claim (as defined in section 4.6); and

- (iv) all other obligations or liabilities disclosed to TISH in a schedule hereto, fully reserved on the Closing Balance Sheet or otherwise specifically disclosed to TISH through the delivery of a written statement prior to the date hereof describing such liability, but in no event any liability described in section 2.4(c), (d), (f) or (g).
- 2.2 Distributions. Transferor and the Stockholders acknowledge that they have instructed TISH to make the cash payments and to cause the TISH Shares to be issued in the manner provided in section 2.1 and TISH bears no responsibility with respect to the further distribution of such payments or TISH Shares.

## 2.3 Working Capital Minimum.

- (a) Within 90 days of the Closing Date, TISH shall cause to be delivered to Transferor a statement of assets and liabilities of Transferor, reflecting the Assets and the Assumed Liabilities (the "Closing Balance Sheet") prepared on an accrual basis in accordance with generally accepted accounting principles as of the close of business on the Closing Date. If the sum of the current liabilities of Transferor shown on the Closing Balance Sheet is greater than the sum of the current assets of Transferor shown on the Closing Balance Sheet, Transferor shall pay in cash to TISH the amount that the sum of the current liabilities exceeds the sum of the current assets (the "Adjustment Amount"), with such payment to be made within 30 days of the final determination of the Adjustment Amount. In the event that the sum of the current assets of Transferor shown on the Closing Balance Sheet is greater than the sum of the current liabilities of Transferor shown on the Closing Balance Sheet, TISH shall pay in cash to Transferor the amount that the sum of the current assets exceeds the sum of the current liabilities (the "Reverse Adjustment Amount") with such payment to be made within 30 days of the final determination of the Reverse Adjustment Amount.
- (b) Within 30 days after the Transferor's receipt of the Closing Balance Sheet, Transferor may deliver to TISH a written statement setting forth any disagreements of Transferor with any of the information contained in the Closing Balance Sheet that affects the determination of the Adjustment Amount (or the Reverse Adjustment Amount). If Transferor does not

submit such a statement within the 30-day period, then the Adjustment Amount (or the Reverse Adjustment Amount), if any, as determined in accordance with the Closing Balance Sheet, shall be final and binding on the parties. If TISH and Transferor do not resolve the disagreements within 15 days after delivery of the written statement by Transferor, a firm of independent certified public accountants, whose decisions on the matter shall be binding and whose fees and expenses shall be borne 50% by Transferor and 50% by TISH, shall be designated by agreement between them; if they fail to agree on the firm to decide the matter

-4-

within an additional 10 days, the accountants shall be selected by the president of the American Institute of Certified Public Accountants. Each party shall allow the other party reasonable access to all information reasonably requested by such other party to prepare the Closing Balance Sheet or the Transferor's statement hereunder, as the case may be.

2.4 Limitation on Assumption of Liabilities. Except as provided in section 2.1(c), TISH shall not assume, and shall not have any liability for, any liability or obligation of Transferor, and Transferor shall pay, perform and discharge all such liabilities and obligations. Without limiting the generality of the preceding sentence, TISH shall not assume or be responsible for (a) any liability or obligation arising out of any claim, litigation or proceeding, or any lease, commitment or other agreement, relating to the operation of the Business or the Real Property (as hereafter defined) or the occurrence of any event or the existence of any condition prior to the Closing Date, except to the extent provided in section 2.1(c), (b) any liability or obligation to any employee of Transferor for compensation or benefits (including vacation and severance pay) incurred or accrued on or prior to the Closing Date, except to the extent provided in section 2.1(c), (c) any deferred compensation obligation or any liability or obligation of Transferor arising out of or in connection with any employee benefit plan maintained by, contributed to or sponsored by an ERISA Affiliate (as defined in section 4.15(a)) or any Plan (as defined in section 4.15(a)), or any other liability or obligations for employee post-retirement life insurance or health care benefits to employees of Transferor who do not become active employees of TISH or an affiliate of TISH, (d) any liability or obligation of Transferor or the Stockholders for taxes of any kind, (e) any payable or other liability to any Stockholder or other affiliate of Transferor (other than accrued payroll obligations incurred in the ordinary course and fully accrued on the Closing Balance Sheet), (f) any indebtedness for borrowed money or other financial indebtedness, including indebtedness to a bank, and any interest or penalties with respect to obligations for borrowed money, (g) legal, accounting, brokerage or finders fees, sales, income or other taxes or other expense of Transferor in connection with this agreement or the transactions contemplated hereby or (h) any other liability or obligation of any kind relating to the operations of the Business, or the occurrence of any event on or before the Closing Date, whether that liability is known or determined as of the Closing Date or unknown or undetermined as of that date, except to the extent provided in section 2.1(c).

- 2.5 Cancellation of Affiliate Payables and Receivables. Immediately prior to the closing, any payable or other liability to, or receivable or other asset due from, a Stockholder, any affiliate of a Stockholder, or any Affiliate will be canceled without any cost or adverse consequence to Transferor.
- 3. Closing. The closing under this agreement shall take place at the offices of Proskauer Rose LLP, 1585 Broadway, New York, New York 10036, on March 15, 2000. The date on which the closing is held is referred to in this agreement as the "Closing Date". At the closing, the parties shall execute and deliver the documents referred to in section 8. If the closing has not occurred on or prior to June 30, 2000, either party may terminate this agreement by notice to the other; upon such termination, neither party shall have any liability arising out of this agreement other than any liability resulting from its breach of or misrepresentation under this agreement prior to termination.

-5-

- 4. Representations and Warranties by Transferor and Stockholders. Transferor and the Stockholders jointly and severally represent and warrant to TISH and TOIC that, as of the date hereof:
- 4.1 Transferor's Organization, Authority and Ownership. Transferor is a corporation duly organized, validly existing and in good standing under the law of the State of \_\_\_\_\_\_ and has the full corporate power and authority to enter into and to perform this agreement and to carry on its business as it currently is being conducted. Transferor is duly qualified to do business and is in good standing as a foreign corporation in each jurisdiction listed on schedule 4.1, which is each jurisdiction in which it owns, leases or operates property or is otherwise required to be qualified, except where the failure to be so qualified would not have a material adverse effect on the Business. The Stockholders own of record and beneficially, in the aggregate, all outstanding capital stock of Transferor, and no other party has any equity or other similar interest in Transferor. Transferor does not have any equity or other similar interests in any entity (other than United States Mutual Association, Inc.).
- 4.2 Authorization of Agreement. The execution, delivery and performance of this agreement by Transferor have been duly authorized by all requisite corporate action of Transferor. This agreement constitutes, and the Escrow Agreement, stockholders agreement and employment agreements referred to in sections 2.1(a)(iii), 6.8 and 6.9 (the "Closing Documents") when executed at the closing will constitute, the valid and binding obligation of Transferor and each Stockholder and is enforceable against Transferor and each Stockholder in accordance with its respective terms, except as enforceability may be limited by bankruptcy, insolvency or other similar laws affecting the enforcement of creditors' rights in general and subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

4.3 Consents of Third Parties. Subject to receipt of the consents and approvals referred to on schedule 4.3, the execution, delivery and performance of this agreement and Closing Documents by Transferor and each Stockholder will not (i) conflict with the certificate of incorporation or by-laws of Transferor; (ii) conflict with, or result in the breach or termination of, or constitute a default under, any agreement, lease, commitment, order or other instrument (including, without limitation, those listed on schedule 4.8), or any order, judgment or decree, to which Transferor or any Stockholder is a party or by which Transferor or any Stockholder is bound; (iii) constitute a violation by Transferor or any Stockholder of any law or regulation applicable to it or him; or (iv) result in the creation of any lien, charge or encumbrance upon any of the properties or assets to be sold to TISH under this agreement. Except as set forth on schedule 4.3, no consent, approval or authorization of, or designation, declaration or filing with, any governmental authority is required on the part of Transferor or any Stockholder or their respective affiliates in connection with the execution, delivery and performance of this agreement. In this regard, the Transferor and each Stockholder have executed and delivered to TISH the affidavit attached to this agreement as Exhibit VI.

-6-

#### 4.4 Title to Assets and Related Matters; Condition of Assets.

(a) Transferor has, and at the closing TISH will receive, valid title to all of the assets intended to be sold to TISH pursuant to this agreement, free and clear of any claim, lien, or encumbrance, except for the lien, if any, of current taxes not yet due and payable, and any other lien disclosed on schedule 4.4(a). The Assets include all of the assets reflected on the balance sheet as of December 31, 1999 referred to in section 4.5, except for the Excluded Assets and items sold, transferred or otherwise disposed of since that balance sheet date in the ordinary course of business and substantially consistent with past practice.

(b) Except for the Excluded Assets, the Assets include all of the assets (tangible and intangible) necessary for, or used in, the conduct of the Business in the manner in which it has been conducted by Transferor and those assets (together with TISH's replacements for the Excluded Assets) will be sufficient to enable TISH to conduct the Business in the manner in which it has been conducted by Transferor. Except for the Excluded Assets, no assets, tangible or intangible, of Transferor or any affiliate of Transferor are being used in the Business that are not being sold and transferred to TISH under this agreement.

(c) Schedule 4.4(c) contains a list of all real properties owned or leased by Transferor with respect to the Business. Transferor enjoys peaceful and undisturbed possession under all real property leases covering premises used in the conduct of the Business, and all such real property leases are valid and subsisting and in full force and effect. Except as set forth on

schedule 4.4(c), all improvements on the real properties used by Transferor in connection with the Business are in accordance in all material respects with all applicable laws, ordinances, regulations and orders, including, but not limited to, those applicable to zoning, environment and the establishment and maintenance of working conditions for labor. All of the buildings and structures leased to or owned by Transferor in connection with the Business are, in all material respects, in good condition and are adequate and suitable for their present uses and purposes, and, except as otherwise disclosed on schedule 4.4(c), the transactions contemplated by this agreement will not adversely affect the right of TISH to use those properties for the same purpose and to the same extent after the Closing Date as they were being used by Transferor prior to the date of this agreement. To Transferor's and Stockholders' knowledge, there are no interior or exterior material structural defects or other material defects in the improvements or any material defect in the plumbing, electrical, mechanical, heating, ventilating or air-conditioning systems or other systems. To Transferor and Stockholder's knowledge, all such systems are, in all material respects, in good working order, and all roofs and basements are, in all material respects, in good condition and free of leaks.

With respect to the real property owned by Transferor, (i) Transferor has delivered to TISH a copy of any title insurance policies and surveys for such property in Transferor's possession, (ii) Transferor has good and marketable fee simple title thereto and owns all improvements (including buildings and other structures) thereon, subject only to liens and encumbrances set forth on schedule 4.4(c) and such other liens as do not materially interfere with the operation of the real property or adversely affect the marketability or value of the real property, (iii) Transferor has full and exclusive right, power and authority to convey fee simple

-7-

title to the properties owned by Transferor. To the Transferor's or the Stockholders' knowledge, except as disclosed on Schedule 4.4(c), there are no: (A) pending or contemplated annexation, condemnation or similar proceedings affecting, or which may affect, the real property, (B) proposed or pending proceedings to change or redefine the zoning of the real property, (C) proposed changes in roads which may adversely affect ingress to or egress from the real property, or (D) proposed or pending special assessments affecting the real property or any penalties or interest due with respect to real estate taxes assessed against the real property that are payable by Transferor as all taxes applicable to the real property have been paid in full, except taxes for the current year, which are not yet due and payable.

- (d) Schedule 4.4(d) contains a complete list of all equipment (including computers and office equipment), furniture, vehicles, and other tangible assets used in the Business that had an original cost for any individual item of more than \$1,000.
- (e) To Transferor's and Stockholders' knowledge, all of the tangible assets to be purchased by TISH under this agreement are, in all

material respects, in good operating condition and in good condition of maintenance and repair, subject to normal wear and tear, and are suitable for continued use in the normal course of business.

- 4.5 Financial Statements. Schedule 4.5 contains financial statements consisting of (i) balance sheets of Transferor as of December 31, 1997 and December 31, 1998 and the related statements of income and cash flows for the 12 month periods then ended, including all related notes thereto [with the report of ]; and (ii) a balance sheet of Transferor as of December 31, 1999 and the related statements of income and cash flows for the 12 month period then ended. Except as set forth on schedule 4.5, all such financial statements have been prepared on an accrual basis or a cash basis, as indicated in such financial statements, and fairly present the financial position (on a cash or accrual basis, as applicable) of the Business as of the dates indicated and the results of operations of the Business for the periods indicated. All such financial statements have been prepared in accordance with Transferor's books and records and show all income and expenses attributable to the Business during the respective periods covered by them. All of Transferor's books of account relating to the Business have been exhibited or made available to TISH, and those books of account accurately record all transactions of the Business during the respective periods covered by them. All of the accounts receivable reflected on the balance sheet as of December 31, 1999 arose from, and all of the accounts receivable as of the Closing Date will have arisen from, bona fide transactions in the ordinary course of business and, except as disclosed on schedule 4.5, none of them is or will be subject to any defense, counterclaim or setoff and all of such accounts receivable shall be collectible within 90 days after such obligations arose in their full recorded amounts, subject to any reserves therefor reflected on the balance sheet as of December 31, 1999 or the Closing Balance Sheet, respectively.
- 4.6 Undisclosed Liabilities. Except to the extent reflected on the balance sheet as of December 31, 1999 referred to in section 4.5 or reflected in the notes to that balance sheet or otherwise disclosed on a schedule to this agreement, Transferor does not, and on the Closing Date will not, have any liability or obligation of any kind relating to the Business,

-8-

whether accrued, absolute, contingent or otherwise, other than (a) liabilities and obligations under orders, commitments, agreements and leases entered into in the ordinary course of business (which, to the extent required by section 4.8, are referred to on schedule 4.8) and (b) other liabilities and obligations that arose in the ordinary course of the Business and are not material in amount. Notwithstanding the foregoing, any liability resulting from a claim asserted by any individual against TISH after the closing with regard to a negative report issued by Transferor in the ordinary course of business prior to the closing shall not be considered an undisclosed liability (but shall instead be deemed an "Assumed Liability" for all purposes of this Agreement), so long as such assertion of a claim was not known by Transferor to exist prior to the closing ("Unknown Report Claim").

- 4.7 Absence of Certain Changes. Since December 31, 1999, Transferor has operated the Business in the ordinary course and consistent with past practices, and, except as set forth on schedule 4.7:
- (a) Transferor has not entered into any transaction or incurred any liability or obligation with respect to the Business that was unusual in nature or amount or was entered into or incurred other than in the ordinary course of business;
- (b) there has not been any material adverse change in the condition (financial or otherwise) or prospects of the Business, or in the assets to be acquired by TISH pursuant to this agreement;
- (c) Transferor has not sold or transferred any assets (other than Excluded Assets) relating to or used in the Business other than in the ordinary course of business and other than assets that have been replaced with other assets of equal or greater value;
- (d) Transferor has not (i) granted or agreed to grant any general increase in any rate or rates of salaries or compensation to employees or agents of the Business or any specific increase in the salary or compensation to any such employee or agent whose total salary and compensation after such increase would be at an annual rate in excess of \$30,000, (ii) increased or amended any benefits of those employees or agents, (iii) amended, adopted or terminated any Plan, or (iv) entered into any employment, bonus, deferred compensation or any other agreement with any employee or agent;
- (e) Transferor has not made any material change in the manner in which the Business has been operated or the accounting principles or practices employed by Transferor in connection with the Business, and no person who performed substantial services with respect to the Business during the twelve months preceding the date of this agreement is currently employed by or performing substantial services for any affiliate of Transferor;
- (f) Transferor has not taken any action to accelerate collection of accounts receivable or to defer payment of liabilities, except in the ordinary course consistent with past practices; and

-9-

- (g) to Transferor's and the Stockholders' knowledge, Transferor has not amended, terminated, canceled or compromised any material claim relating to the Business, or waived or allowed to lapse or terminate any other rights of substantial value to the Business.
- 4.8 Lists of Agreements, etc. Schedule 4.8 contains, with respect to the Business, a true and complete list of: (a) all commitments and agreements for the purchase of materials or supplies or the receipt of services that involve an expenditure by Transferor of more than \$10,000 for any one commitment

or two or more related commitments; (b) all agreements relating to pre-employment investigation and screening services; (c) all leases or other rental agreements under which Transferor is lessee or lessor; (d) all license agreements under which Transferor is either licensee or licensor; (e) all collective bargaining agreements and all employ ment and consulting agreements and agreements with independent contractors that provide for compensation in excess of \$10,000 a year; (f) all agreements between Transferor and any affiliate of Transferor (or any entity in which any of them has a material interest); (g) all commitments and other agreements limiting Transferor's freedom to engage in any line of business or to compete with any other person; and (h) all other agreements, commitments and orders to which Transferor is a party or by which it is bound that involve more than \$10,000 or are otherwise material to the Business or the Assets. True and complete copies of the agreements, commitments and leases referred to on schedule 4.8 (if in written form and summaries thereof if in oral form) have been delivered or made available to TISH .

4.9 Status of Agreements. Except as set forth on schedule 4.9, all leases, commitments and other agreements of Transferor were entered into in connection with and in the ordinary course of the Business and on an arm's length basis. Each of the agreements, commitments, orders and leases referred to in section 4.8 is presently in full force and effect in accordance with its terms and Transferor is not in default, and, to the knowledge of Transferor and the Stockholders, no other party is in default, under any agreement, commitment, order or lease referred to in section 4.8 and each of those agreements, commitments, orders and leases is valid and binding upon and enforceable against each of the parties thereto in accordance with its terms, except as enforceability may be limited by bankruptcy, insolvency or other similar laws affecting the enforcement of creditors' rights in general. To Seller's and Stockholder's knowledge, no party to any of the agreements, commitments, orders or leases referred to in section 4.8 has made, asserted in writing or has any defense, setoff or counterclaim under any of those agreements, commitments, orders or leases or has exercised any option granted to it to cancel or terminate its agreement, to shorten the term of its agreement, or to renew or extend the term of its agreement, and Transferor has not received any written notice to that effect.

4.10 Litigation; Compliance with Laws. Except as set forth on schedule 4.10, within the past three years there has not been, and there is not currently, any claim, litigation, proceeding or governmental investigation pending or, to the knowledge of Transferor or the Stockholders, threatened, or any order, injunction or decree outstanding against or relating to Transferor with respect to the Business or any of the Assets. Subject to the last sentence of section 4.6, neither Transferor nor the Stockholders know of any basis for future claims, litigations, proceedings or investigations against Transferor with respect to the Business or any of the Assets. Transferor is not in violation, in any material respect, of any law, regulation,

ordinance, or any other requirement of any governmental body or court, including the Fair Credit Reporting Act, arising out of the operation of the Business (including, without limitation, the use of the databases and any information contained therein and related software) and no written notice has been received by Transferor or any of its officers, directors or stockholders alleging any such violation.

4.11 Intellectual Property. Schedule 4.11 contains a true and complete list of the trademarks, trade names, copyrights, patents, trade secrets, logos, computer software, databases, web sites, domain names and e-mail addresses (collectively, "Intellectual Property"), used by Transferor in the Business. Schedule 4.11 also contains a true and complete list of all patent, copyright, trademark, database and domain name registrations or registration applications that have been obtained or filed by Transferor anywhere in the world. Except as set forth on schedule 4.11, Transferor owns or possesses adequate licenses or other rights to use , free and clear of any claims, liens or encumbrances, each of the items of Intellectual Property listed on schedule 4.11 (including, without limitation, any information contained in the databases), and they constitute all the Intellectual Property necessary for the continued operation of the Business in a manner consistent with past practices. Schedule 4.11 identifies which items of Intellectual Property are owned by Transferor and which items are used by Transferor under license. Except as set forth on schedule 4.11, (a) to the knowledge of Transferor and the Stockholders (i) there is no violation by others of any right of Transferor with respect to its Intellectual Property, and (ii) Transferor is not, in connection with the Business, infringing upon any Intellectual Property or other rights of any third party, including, without limitation, by Transferor's use of the Intellectual Property; (b) no proceedings are pending or threatened, and no claim has been received by Transferor or the Stockholders alleging any such violation; (c) no royalty or similar fee of any kind is payable by Transferor for the use of any of the Intellectual Property; and (d) Transferor has not granted any person or entity any interest, as licensee or otherwise, in or to any one or more items of the Intellectual Property.

4.12 Labor Matters. Except as set forth on schedule 4.12, with respect to the Business (a) Transferor is in compliance, in all material respects, with all applicable laws respecting employment and employment practices, terms and conditions of employment, and wages and hours, and is not engaged in any unfair labor practice; (b) there is no unfair labor practice charge or complaint against Transferor pending before the National Labor Relations Board, any state labor relations board or any court or tribunal and, to the knowledge of Transferor, none is or has been threatened in writing; (c) there is no labor strike, dispute, request for representation, slowdown or stoppage actually pending against or affecting Transferor and, to the knowledge of Transferor and the Stockholders, none is or has been threatened in writing; (d) no grievance which might have a material adverse effect on the conduct of the Business or any arbitration proceeding arising out of or under any collective bargaining agreement is pending and, to the knowledge of Transferor and the Stockholders, none is or has been threatened in writing; and (e) there is no workman compensation or similar claim pending or threatened. Except as set forth on schedule 4.12, to the knowledge of Transferor and the Stockholders, there has been no sexual harassment or similar claim against any employee of the Business and no employee of the Business has at any time within the three years preceding the date of this agreement asserted in writing against Transferor or any of its employees any claim of discrimination.

-11-

- 4.13 Environmental Matters. With respect to the ownership and operation of the Business and any currently or formerly owned, operated or used real property, to the knowledge of the Stockholders, except as set forth on schedule 4.13:
- (a) Transferor and all of the real property owned, leased or used by Transferor (the "Real Property") is in compliance in all material respects with all federal, state and local laws and regulations relating to injury to or pollution or protection of human health or the environment (the "Environmental Laws"), including the common law, environmental permits and laws and regulations relating to emissions, discharges, releases or threatened releases of chemicals, pollutants, contaminants, wastes, substances, petroleum and petroleum products (collectively, "Materials of Environmental Concern"), or otherwise relating to the manufacture, generation, processing, distribution, use, treatment, storage, disposal, transport or handling of Materials of Environmental Concern;
- (b) there are no past or present actions, activities, circumstances, conditions, events or incidents, including, but not limited to, the release, emission, discharge or disposal or threatened release of any Material of Environmental Concern, that could form the basis of any material claim against Transferor or the Real Property or could reasonably be expected to constitute a material violation by Transferor (or, after the closing, TISH) or is reasonably likely to require investigation, remediation or monitoring under the Environmental Laws, and neither Transferor nor any Stockholder has received any written notice of actual or potential liability under the Environmental Laws; and
- (c) there are no underground storage tanks located on the Real Property, there are no friable asbestos-containing materials in or on the Real Property and no polychlorinated biphenyls (PCBs) are used or stored by Transferor at the Real Property.

#### 4.14 ERISA.

(a) Schedule 4.14 contains a true and complete list of all "employee benefit plans," within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and any other bonus, profit sharing, pension, severance, savings, deferred compensation, fringe benefit, insurance, welfare, post-retirement benefit, health, life, stock option, stock purchase, restricted stock, tuition refund, service award, company car, scholarship, relocation, disability, accident, sick, vacation, individual

employment, consulting, compensation, incentive, commission, payroll practices, retention, change in control, noncompetition, or other plan, agreement, policy, trust fund, or arrangement (whether written or unwritten, insured or self-insured) established, maintained, sponsored, or contributed to (or with respect to which any obligation has been undertaken) by Transferor or any entity that would be deemed a "single employer" with the Transferor under section 414(b), (c) (m) or (o) of the Internal Revenue Code of 1986, as amended (the "Code") or section 4001 of ERISA (an "ERISA Affiliate") on behalf of any employee, consultant, director or shareholder of Transferor (whether current, former or retired) or their beneficiaries (each a "Plan" and, collectively, the "Plans"). With respect to each Plan, to the extent applicable, true and complete copies of the plan and documents (including amendments) and summary plan description, IRS Form 5500, if any, IRS

-12-

determination letter and other material documents relating to the Plan have been delivered to TISH. None of the Company, any ERISA Affiliate, or any of their respective predecessors has ever contributed to, contributes to, has ever been required to contribute to, or otherwise participated in or participates in or in any way, directly or indirectly, has or could reasonably be expected to have any liability with respect to any plan subject to Section 412 of the Code, Section 302 of ERISA or Title IV of ERISA, including, without limitation any, "multiemployer plan" (within the meaning of Sections (3)(37) or 4001(a)(3) of ERISA or Section 414(f) of the Code), or any single employer pension plan.

(b) Each Plan complies in all material respects and has been maintained and operated in all material respects in compliance with its terms and applicable law, including, without limitation, ERISA and the Code. With respect to each of the Plans, except as set forth on schedule 4.14: (i) each Plan intended to qualify under Section 401(a) of the Code is qualified and has received a determination letter from the IRS to the effect that the Plan is qualified under Section 401 of the Code and any trust maintained pursuant thereto is exempt from federal income taxation under Section 501 of the Code and nothing has occurred or is expected to occur through the Closing Date that caused or could cause the loss of such qualification or exemption or the imposition of any penalty or tax liability; (ii) all payments required by any Plan, any collective bargaining agreement or other agreement, or by law with respect to all periods through the closing shall have been made prior to the Closing Date to the extent due and payable on the Closing Date (on a pro rata basis where such payments are otherwise discretionary at year end) or provided for by the Company as applicable, by full accruals as if all targets required by such Plan had been or will be met at maximum levels) on its financial statements; (iii) no claim, lawsuit, arbitration or other action has been threatened, asserted or instituted against the Plans (other than non-material routine claims for benefits, and appeals of such claims), any trustee or fiduciaries thereof, the Company, any ERISA Affiliate, any director, officer, or employee thereof, or any of the assets of any trust of the Plans and the Company is unaware of any facts that could reasonably be expected to give rise to any such claim, lawsuit, arbitration, or other action; (iv) no "prohibited

transaction," within the meaning of Section 4975 of the Code and Section 406 of ERISA, has occurred or is expected to occur with respect to the Plan; (v) no Plan is under audit or investigation by the IRS, U.S. Department of Labor, or any other governmental authority and no such completed audit, if any, has resulted in the imposition of any tax or penalty and (vi) with respect to each Plan that is funded mostly or partially through an insurance policy, neither the Company nor any ERISA Affiliate has any liability in the nature of retroactive rate adjustment, loss sharing arrangement or other actual or contingent liability arising wholly or partially out of events occurring on or before the Closing Date.

(c) Except as provided in schedule 4.14(c), the consummation of the transactions contemplated by this agreement will not give rise to any liability including, without limitation, liability for severance pay, unemployment compensation, termination pay, or withdrawal liability, or accelerate the time of payment or vesting or increase the amount of compensation or benefits due to any employee, director or shareholder of Transferor (whether current, former, or retired) or their beneficiaries solely by reason of such transaction or by reason of a termination of employment following such transaction. No amounts payable under any Plan

-13-

will fail to be deductible for federal income tax purposes by virtue of Sections 280G or 162(m) of the Code.

- (d) Neither the Company nor any ERISA Affiliate maintains, contributes to, or in any way provides for any benefits of any kind whatsoever (other than under Section 4980B of the Code, the Federal Social Security Act, or a plan qualified under Section 401(a) of the Code) to any current or future retiree or terminee. No event, condition, or circumstance exists that could result in an increase of the benefits provided under any Plan based on the Plan documents from the level of benefits for the most recent fiscal year ended before the Closing Date; and none of Transferor, any ERISA Affiliate or officer or employee thereof, has made any legally binding promises or commitments to create any additional plan, agreement or arrangement, or to modify or change any existing Plan. Neither the Company nor any ERISA Affiliate has any unfunded liabilities pursuant to any Plan that is not intended to be qualified under Section 401(a) of the Code. No event, condition, or circumstance exists that could reasonably be expected to prevent the amendment or termination of any Plan.
- 4.15 Brokers and Finders. Neither Transferor nor any Stockholder has employed any broker or finder or incurred any liability (on its behalf or on behalf of others, including TISH) for any investment banking fees, brokerage fees or commissions or finders fees in connection with the transactions contemplated by this agreement.
- 4.16 Transactions with Affiliates. Except as set forth on schedule 4.16, Transferor is not, and since December 31, 1998, has not, engaged,

with respect to the Business, in any transaction with any officer or director of Transferor, any of their family members, or any entity in which any of them has an interest.

- 4.17 Permits and Licenses. Transferor has, and is in compliance, in all material respects, with, all permits, licences, franchises and other authorizations ("Licenses") necessary for the conduct of the Business, and all such Licenses are valid and in full force and effect. All Licenses held by Transferor that are material to the Business are listed on schedule 4.17.
- 4.18 Taxes. Transferor has timely filed all federal, state and other tax returns required by law to be filed by it, and each of those tax returns is correct and complete in all respects. Transferor has paid, or made provision in its financial statements referred to in section 4.5 for payment of, all of its taxes accrued through the date of those financial statements, except that any accrued but unpaid taxes for 1999 and beyond are not reflected in financial statements prepared on a cash basis. There are no claims pending or threatened in writing against Transferor for past due taxes. All taxes that are or were required by law to be withheld or collected by Transferor have been duly withheld or collected and paid to the proper tax authority. Since its incorporation, Transferor has been an S corporation within the meaning of the Code.

4.19 Insurance.

-14-

- (a) Schedule 4.19(a) contains a complete list of all insurance policies relating to the Business under which Transferor is currently an insured, a named insured or otherwise the principal beneficiary of coverage. True and complete copies of all of those policies have been delivered or made available to TISH. Transferor has not received any notice that any of those insurance policies are not in full force and effect. Transferor is not in breach or default (including any breach or default with respect to the payment of premiums or the giving of notice), and no event has occurred which, with notice or the lapse of time, would constitute such a breach or default or permit termination or modification, under any those insurance policies. Transferor has not received any written notice of cancellation or intent to cancel any of those insurance policies. Transferor has paid all premiums due and payable with respect to those insurance policies.
- (b) Schedule 4.19(b) sets forth a list of all claims of Transferor relating to the Business that are pending with any insurance company.
  - 4.20 Employees.
- (a) Schedule 4.20 is a true and complete list of all current employees of Transferor or any of its affiliates who perform, or who at any time during the twelve months preceding the date of this agreement performed, substantial services with respect to the Business, which list identifies the

name and position of each such employee, and the following compensation information for fiscal years 1997, 1998 and 1999: (a) annual base salary, (b) annual bonus, (c) commissions, (d) perquisites, (e) severance, (f) all other items of compensation and (g) date of birth and date of hire.

- (b) Except as set forth on schedule 4.14, Transferor does not have any severance policy and no employee of the Business is entitled to any severance payment, either by law or by agreement, upon the termination of his or her employment. No employee of the Business is represented by any union or other collective bargaining agent and there are no collective bargaining or other labor agreements with respect to those employees.
- (c) Except as set forth on schedule 4.20, no employee has threatened in writing, nor to the knowledge of Transferor or the Stockholders, intends to terminate his or her employment.
- 4.21 Year 2000 Compliance. Except as set forth on schedule 4.21, the computer systems (including all work stations and other components) used in the Business both before and after the beginning of the year 2000, (i) are capable of recognizing, processing, managing, representing, interpreting, and manipulating correctly date related data for dates earlier and later than January 1, 2000, including, but not limited to, calculating, comparing, sorting, storing, tagging, and sequencing, without resulting in or causing logical or mathematical errors or inconsistencies in any user-interface functionalities, data storage, data fields, calculations, reports, processing, or any other input or output, (ii) have the ability to provide date recognition for any data element represented without a date, or whose year is represented by only two digits and the ability to automatically function into and beyond the year 2000 without human

-15-

intervention, (iii) correctly interpret data, dates and time into and beyond the year 2000, (iv) have the ability not to produce non-compliance in existing information, nor otherwise corrupt such data, and (v) have the ability to successfully interface with internal and external applications or systems that have not yet achieved year 2000 compliance during the time in which the systems and such applications and systems co-exist.

4.22 Customers. Schedule 4.22 contains a complete list of the twenty largest customers (measured by dollar volume in each case) of Transferor during each of the last three fiscal years and during the first month of the current fiscal year, showing, with respect to each, the name and address, dollar volume involved, and the percentage of each of Transferor's lines of business which each such customer represented. The Transferor is not required to provide any bonding or other financial security arrangements in connection with any transactions with any customers in the ordinary course of its business. Except as set forth on schedule 4.22, Transferor is not and within the last two years has not been engaged in any material disputes with its customers. To Transferor's and the Stockholders' knowledge, except as set forth on schedule

- 4.22, there has been no loss or threatened loss of any material customer nor is Transferor nor the Stockholders aware of any material customer which is considering termination, nonrenewal or any adverse modification of its arrangements with the Transferor before the closing or with TISH after the closing.
- 4.23 No Misrepresentation. No representation or warranty by Transferor and the Stockholders in this agreement (including the schedules and exhibits to this agreement) contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in this agreement (including the schedules and exhibits to this agreement) not misleading.
- 5. Representations and Warranties by TISH. TISH and TOIC jointly and severally represent and warrant to Transferor and the Stockholders that, as of the date hereof:
- 5.1 TISH 's Organization. TISH is a corporation duly organized, validly existing and in good standing under the law of the State of Delaware and has full corporate power and authority to enter into and to perform this agreement. TISH is a wholly owned subsidiary of TOIC and was formed for the purpose of consummating the transactions contemplated under this agreement and the Affiliate Acquisition Agreements and has conducted no other business prior to the date hereof. TISH has delivered to Transferor a true and correct copy of the certificate of incorporation and bylaws of TISH. The capitalization of TISH immediately after the closing is set forth on schedule 5.1.
- 5.2 Authorization of Agreement. The execution, delivery and performance of this agreement and the Closing Documents by TISH and TOIC have been duly authorized by all requisite corporate action of TISH and TOIC. This agreement constitutes, and the Closing Documents when executed will constitute, the valid and binding obligation of TISH and TOIC enforceable against each of them in accordance with its terms, except as enforceability may be limited by bankruptcy, insolvency or other similar laws affecting the enforcement of creditors' rights in general and subject to general principles of equity (regardless of whether such enforceability is considered in a proceeding in equity or at law).

-16-

5.3 Consents of Third Parties. The execution, delivery and performance of this agreement by TISH or TOIC will not (i) conflict with the certificate of incorporation or by-laws of TISH or TOIC, (ii) conflict with, result in the breach or termination of, or constitute a default under, any lease, agreement, commitment or other instrument, or any order, judgment or decree, to which TISH or TOIC is a party by which it is bound, or (iii) constitute a violation by TISH or TOIC of any law or regulation applicable to it. No consent, approval or authorization of, or designation, declaration or filing with, any governmental authority is required on the part of TISH or TOIC in connection with the execution, delivery and performance of this agreement.

5.4 Brokers and Finders. Neither TISH nor TOIC has employed any broker or finder or incurred any liability (on its behalf or on behalf of others, including Transferor) for any investment banking fees, brokerage fees or commissions or finders fees in connection with the transactions contemplated by this agreement, except for Veronis, Suhler & Associates, Inc. (the fees and costs of which shall be paid by TISH or TOIC).

## 5.5 TISI.

- (a) Simultaneously with the closing, TOIC shall transfer to TISH all of the outstanding shares of capital stock of TISI.
- (b) TISI is a corporation duly organized, validly existing and in good standing under the law of the State of Oklahoma and has the full corporate power and authority to carry on its business as it currently is being conducted.
- (c) Schedule 5.5(c) contains financial statements consisting of (i) balance sheets of TISI as of December 31, 1997 and December 31, 1998, and the related statements of income and cash flows for the 12 month periods then ended, including all related notes thereto, and (ii) a proforma balance sheet of TISI as of December 31, 1999 and the related proforma statement of income and cash flow for the 12 month period then ended (after giving effect to pending and completed acquisitions). Except as set forth on schedule 5.5(c), all such financial statements have been prepared in accordance with generally accepted accounting principles applied on a consistent basis and fairly present the financial position of TISI as of the dates indicated and the results of operations of TISI for the periods indicated. All such financial statements have been prepared in accordance with TISI's books and records and show all income and expenses attributable to TISI's business during the respective periods covered by them. Except as set forth on schedule 5.5(c), as of December 31, 1999, TISI had not incurred any material liability other than as disclosed or reserved for on its proforma financial statements as of December 31, 1999 or in connection with contracts and other commitments, obligations and liabilities entered into or incurred in the ordinary course of business. Since December 31, 1999, there has been no material adverse change in the financial condition of TISI.
- (d) Except as set forth on schedule 5.5(d), there is no claim, litigation, proceeding or governmental investigation pending or, to the best of the knowledge of TISH, threatened, or any order, injunction or decree outstanding against or relating to TISI or TISH which will or is likely to have a material adverse effect on TISI, TISH and their subsidiaries

-17-

taken as a whole. TISH does not know of any basis for future claims, litigations, proceedings or investigations against TISI or TISH which is likely to have a material adverse effect on TISI, TISH and their subsidiaries taken as

a whole. Neither TISI nor TISH is in violation, in any material respect, of any law, regulation, ordinance, or any other requirement of any governmental body or court, including the Fair Credit Reporting Act, arising out of the operation of its business, and no notice has been received by TISI or TISH or any of their respective officers or directors alleging any such violation, in any case where such violation or alleged violation is likely to have a material adverse effect on TISI, TISH and their subsidiaries taken as a whole.

- (e) At the closing, the TISH Shares to be issued will be duly authorized and validly issued fully paid and nonassessable and free and clear of all claims, liens and encumbrances, and the percentage ownership of each of the shareholders shall be as provided on schedule 5.1.
- 5.6 Year 2000 Compliance. The computer systems (including all work stations and other components) used in TISI's business, before and after the beginning of the year 2000, (i) are capable of recognizing, processing, managing, representing, interpreting, and manipulating correctly date related data for dates earlier and later than January 1, 2000, including, but not limited to, calculating, comparing, sorting, storing, tagging, and sequencing, without resulting in or causing logical or mathematical errors or inconsistencies in any user-interface functionalities, data storage, data fields, calculations, reports, processing, or any other input or output, (ii) have the ability to provide date recognition for any data element represented without a date, or whose year is represented by only two digits and the ability to automatically function into and beyond the year 2000 without human intervention, (iii) correctly interpret data, dates and time into and beyond the year 2000, (iv) have the ability not to produce non-compliance in existing information, nor otherwise corrupt such data, and (v) have the ability to successfully interface with internal and external applications or systems that have not yet achieved year 2000 compliance during the time in which the systems and such applications and systems co-exist.
- 5.7 No Misrepresentation. No representation or warranty by TISH or TOIC in this agreement (including the schedules and exhibits to this agreement) contains any untrue statement of a material fact or omits to state a material fact necessary to make the statements contained in this agreement (including the schedules and exhibits to this agreement) not misleading.
- 6. Further Agreements of the Parties; Nondisclosure and Confidentiality.
- 6.1 Access to Information. Prior to the closing, TISH and its representatives may, upon reasonable request during normal business hours, make such investigation of the Real Property, assets and operations of the Business as it may desire, and Transferor shall give to TISH and to its counsel, accountants and other representatives, upon reasonable notice, full access during normal business hours throughout the period prior to the closing to all of the Real Property, assets, books, commitments, agreements, records and files of Transferor relating to the Business and the Real Property and Transferor shall furnish to TISH during that period all

documents and copies of documents and information concerning the Business and the Real Property as TISH reasonably may request. TISH shall hold, and shall cause its representatives to hold, all such information and documents and all other information and documents delivered pursuant to this agreement confidential and, if the transactions contemplated by this agreement are not consummated for any reason, shall return to Transferor all such information and documents and any copies as soon as practicable, and shall not disclose or use any such information (that has not previously been disclosed by a party other than TISH ) to any third party unless required to do so pursuant to a request or order under applicable laws and regulations or pursuant to a subpoena or other legal process. TISH 's obligations under this section shall survive the termination of this agreement.

- 6.2 Conduct of the Business Pending the Closing. Until the closing, Transferor and the Stockholders shall operate the Business in the ordinary course in a manner consistent with past practices and:
- (a) Transferor shall promptly notify TISH in writing of, and furnish any information that TISH reasonably may request with respect to, (i) any claim, litigation, proceeding or governmental investigation threatened in writing by or against Transferor relating to the Business or any material development with respect to any such claim, litigation, proceeding or governmental investigation, (ii) the occurrence of any event or the existence of any state of facts that would result in any of the Transferor's representations and warranties not being true as of the Closing Date, and (iii) any other occurrence of any kind materially adversely affecting the Business or the Assets;
- (b) Transferor shall not (i) grant or agree to grant any general increase in the rates of salaries or compensation of its employees, or any specific increase to any such employee whose total salary or compensation after the increase would be at an annual rate in excess of \$30,000, or any increase or amend any benefits of employees or agents of the Business (including, without limitation, increase in the pension, retirement or other employment benefits of the employees of the Business, (ii) amend, adopt or terminate any Plans or (iii) enter into any employment, bonus, deferred compensation or any other agreement with any employee or agent;
- (c) Transferor shall use reasonable efforts, consistent with its past practices, (i) to preserve the business organization of the Business intact and to preserve the goodwill and business of those having business relations with the Business, (ii) to retain the services of the employees of the Business, and (iii) to preserve all trademarks, trade names, logos and copyrights and related registrations of the Business;
- (d) Transferor shall not (i) enter into or renew any agreement, commitment or lease which, if entered into prior to the date of this

agreement, would have been required to be included in schedule 4.8, (ii) cause or take any action to allow any lease, agreement or commitment relating to the Business to lapse (other than in accordance with its terms), to be modified in any material adverse respect, or otherwise to become impaired in any manner, (iii) amend, terminate,

-19-

cancel or compromise any material claim relating to the Business, or waive or allow to lapse or terminate any other rights of substantial value to the Business, or (iv) make or commit to make any capital expenditures which, individually or in the aggregate, involve payments in excess of \$50,000;

- (e) except in the ordinary course and substantially consistent with past practice, Transferor shall not (i) enter into any transaction or voluntarily incur any liability or obligation that is material to the Business or (ii) sell or transfer any of the assets used in the Business, other than assets that have worn out or been replaced with other assets (other than Excluded Assets) of equal or greater value or assets that are no longer needed in the operation of the Business;
- (f) Transferor shall not make any material change in the manner in which the Business is operated or the accounting principles or practices employed in connection with the Business;
- (g) Transferor shall duly comply, in all material respects, with all laws, ordinances, orders, injunctions and decrees applicable to the operation of the Business;
- (h) Transferor shall maintain all of the tangible assets used in the Business in customary repair, maintenance and condition, except to the extent of normal wear and tear, and Transferor shall replace any items of equipment at time intervals consistent with past practices;
- (i) Transferor shall maintain insurance on the tangible assets used in the Business and on the Business in accordance with past practices;
- (j) Transferor shall not take any action to accelerate collection of accounts receivable or to defer payment of liabilities, except in the ordinary course consistent with past practices; and
- (k) Transferor may make cash distributions to its Stockholders provided such distributions do not result, as of the closing, in current liabilities exceeding current assets.
- 6.3 Other Action. No party to this agreement shall take any action that would result in any of its representations and warranties not being true as of the Closing Date. Each of the parties to this agreement shall use its best efforts to cause the fulfillment at the earliest practicable date of all of the

conditions to the obligations of the parties to consummate the transactions under this agreement.

6.4 Consents. Transferor and the Stockholders shall use their best efforts to obtain at the earliest practicable date, in form and substance reasonably satisfactory to TISH, all consents and approvals required to assign to TISH the rights of Transferor under any agreement, commitment or order that requires consent to assignment (including, but not limited to, those set

-20-

forth on schedule 4.3), without any condition adverse to TISH or the operation of the Business after the Closing Date, but if any consent is not obtained (and, accordingly, pursuant to section 1.2 is excluded from the sale pursuant to this agreement), Transferor and the Stockholders shall use their best efforts to keep the agreement in effect and to give TISH the benefit of the agreement to the same extent as if it had been assigned to TISH, and TISH shall perform the obligations under the agreement relating to the benefit obtained by TISH. Nothing in this agreement shall be construed as an attempt to assign any agreement or other instrument that by its terms is nonassignable without the consent of the other party.

- 6.5 Expenses. TISH and Transferor shall bear their own respective expenses incurred in connection with this agreement and in connection with all obligations required to be performed by each of them under this agreement.
- 6.6 Sales Taxes. Transferor shall pay any sales or transfer taxes payable in connection with the sale of the Assets pursuant to this agreement.
  - 6.7 Covenants Against Competition, Solicitation and Disclosure.
- (a) To accord to TISH the full value of its acquisition, during the Noncompete Term, as defined below, Transferor and the Stockholders shall not, directly or indirectly, engage, or be interested in (as owner, stockholder, partner, member, manager, lender, agent or otherwise) or provide any services to any business or entity that engages, anywhere in the United States, in the operation of pre-employment investigation and screening services, except pursuant to any employment, consulting or transition services agreements entered into between the Stockholders and TISH.
- (b) During the Noncompete Term, Transferor and the Stockholders shall not, directly or indirectly, employ or solicit for employment or consulting, on its own behalf or on behalf of any other person or entity, or otherwise encourage the resignation of, any employee of Transferor who provided substantial services to the Business at any time during the one year period ending on the Closing Date.
- (c) Transferor and the Stockholders shall not at any time hereafter disclose to anyone, or use in competition with the Business, any information relating to the Business, which is known or becomes known to

Transferor or the Stockholders, including, without limitation, trade secrets, know-how, financial statements and data, contracts (whether oral or written), customer and advertiser lists, rate schedules, pricing policies, marketing plans and strategies, and business acquisition plans ("Confidential Information") [and shall not provide to anyone in competition with the Business any information derived from Transferor's operation of the Excluded Business]. Confidential Information shall not include information which Transferor or the Stockholders can demonstrate has become generally available to the public other than as a result of a disclosure by Transferor or the Stockholders.

(d) Transferor and the Stockholders acknowledge that the remedy at law for breach of the provisions of this section 6.7 will be inadequate and that, in addition to any

-21-

other remedy TISH may have, it shall be entitled to an injunction restraining any breach or threatened breach, without any bond or other security being required and without the necessity of showing actual damages. If any court construes any covenant in this section 6.7 to be unenforceable in any respect, the court may reduce the duration or area to the extent necessary so that the provision is enforceable, and the provision, as reduced, shall then be enforceable.

- (e) "Noncompete Term" shall mean the later of the three year period after the Closing Date or, with respect to each Stockholder, the three year period after the termination of such Stockholder's employment with TISH pursuant to an employment agreement referred to in section 6.9.
- 6.8 Stockholders' Agreement. At the closing, TISH, TOIC and the Stockholders shall execute and deliver a stockholders' agreement in the form of exhibit VII.
- 6.9 Employment Agreements. At the closing, TISH and the Stockholders shall execute and deliver employment agreements in the form of exhibit VIII.
- 6.10 Escrow Agreement. At the closing, TISH and the Stockholders shall execute and deliver the Escrow Agreement.
- 6.11 Transfer of TISI. Simultaneously or prior to the closing, TOIC shall transfer all of the outstanding shares of TISI to TISH and ensure the completion of all the actions and transactions described in schedule 6.11.
- [6.12 License. Transferor and the Stockholders hereby agree to provide to TISH periodically (but in no event less frequently than monthly) all incident or theft information arising in connection with the Excluded Business (the "Licensed Data") and hereby grant to TISH a perpetual, non-exclusive, royalty free, worldwide license to use the Licensed Data in connection with the Business, but only to the extent that Transferor and/or the Stockholders have

the right to so license all or a portion of the Licensed Data. During the Noncompete Term, Transferor and the Stockholders agree not to use, or license to any other party to use, the Licensed Data to a third party in connection with any business competitive with the Business. Transferor and the Stockholders will use commercially reasonable efforts to obtain the consent of the owner of any Licensed Data (to the extent Transferor or the Stockholders are not the owner) to allow such data to be licensed to TISH pursuant hereto.]

6.13 Interim Financial Statements. Transferor shall promptly (but, in any event, not later than the date such statements customarily are completed by Transferor) deliver to TISH copies of any monthly or quarterly financial statements or other reports relating to the Business that may be prepared by Transferor during the period from the date of this agreement to the Closing Date. All such financial statements shall be prepared from the books and records of the Business and shall show all income and expenses attributable to the Business and shall fairly present the financial position and results of operations of the Business as of and for the periods indicated. Transferor shall also furnish to TISH any other information concerning the financial and operating condition of the Business as TISH from time to time reasonably may request.

-22-

- 6.14 Further Assurances. At any time and from time to time after the Closing Date each party shall, without further consideration, execute and deliver to the other such other instruments of transfer and assumption and shall take such other action as the other may reasonably request to carry out the transfer of the operations, rights and assets and assumption of liabilities contemplated by this agreement.
- 6.15 Post-Closing Services. For a period of [six] months after the closing, each of the Stockholders who is not otherwise entering into employment or consulting agreements with TISH at the closing agrees to continue to work for the Business on a basis and in a manner consistent with past practices with respect to that Stockholder, in order to transfer the Business to TISH (without any obligation to incur expenses), subject to the supervision and instructions of the officers of TISH, and such Stockholders shall receive the same salary and benefits as they had received immediately prior to the date of this agreement [as described on schedule 6.15]. During such period, the Stockholders who are entering into employment or consulting agreements with TISH at the closing agree to comply with the provisions of such employment and consulting agreements. At TISH's option, the Stockholders' services under this agreement may be terminated at any time on 5 business days' prior notice, subject, however, to the terms of any employment or consulting agreements.

## 7. Conditions to Closing.

7.1 Conditions Precedent to Obligations of TISH and TOIC. TISH's and TOIC's obligation to consummate the transactions under this agreement is subject to the fulfillment, prior to or at the closing, of each of the following

conditions (any or all of which may be waived by TISH and TOIC):

- (a) all representations and warranties of Transferor and the Stockholders to TISH and TOIC shall be true and correct in every material respect as of the time of the closing, with the same effect as though those representations and warranties had been made again at and as of that time;
- (b) Transferor and the Stockholders shall have performed and complied with all obligations and covenants required by this agreement to be performed or complied with by them prior to or at the closing;
- (c) TISH and TOIC shall have been furnished with a certificate (dated the Closing Date and in form and substance reasonably satisfactory to TISH) executed by an officer of Transferor, certifying to the fulfillment of the conditions specified in sections 7.1(a) and 7.1(b);
- (d) the transactions contemplated by each Affiliate Acquisition Agreement shall be consummated at the closing;
- (e) TISH and TOIC shall have been furnished with an opinion of Jenner & Block, counsel to Transferor and the Stockholders, in the form of exhibit IX;

-23-

- (f) Transferor and the Stockholders shall have duly received, without any condition adverse to the Business, TISH or TOIC, all consents and approvals listed on schedule 4.3, each in form and substance reasonably satisfactory to TISH;
- (g) there shall not be in effect any injunction or restraining order issued by a court of competent jurisdiction in an action or proceeding against the consummation of the transactions pursuant to this agreement; and
- (h) Transferor or the Stockholders, as the case may be, shall have duly executed and delivered to TISH the agreements referred to in sections 2.1(a) (Escrow Agreement), 6.8 (stockholders' agreement) and 6.9 (employment agreements), and such agreements shall be in full force and effect;
- (i) Merchants Security Exchange ("MSE") shall have duly authorized the sale by United States Mutual Association, Inc. contemplated by the Affiliate Acquisition Agreement relating thereto; and
- $\,$  (j) MSE and TISH shall have entered into a services agreement on terms acceptable to TISH .
- 7.2 Conditions Precedent to Obligations of Transferor and the Stockholders. Transferor's and the Stockholders' obligations to consummate the transactions under this agreement is subject to the fulfillment, prior to or at

the closing, of each of the following conditions (any or all of which may be waived by Transferor or the Stockholders):

- (a) all representations and warranties of TISH and TOIC shall be true and correct in every material respect as of the time of the closing, with the same effect as though those representations and warranties had been made again at and as of that time;
- (b) TISH and TOIC shall have performed and complied with all obligations and covenants required by this agreement to be performed or complied with by them prior to or at the closing;
- (c) Transferor and the Stockholders shall have been furnished with a certificate (dated the Closing Date and in form and substance reasonably satisfactory to Transferor or the Stockholders) executed by an officer of TISH, certifying to the fulfillment of the conditions specified in sections 7.2(a) and 7.2(b);
- (d) the transactions contemplated by each Affiliate Acquisition Agreement shall be consummated at the closing;
- (e) Transferor and the Stockholders shall have been furnished with an opinion of Brian A. Meyer, Esq., general counsel to TISH , in the form of exhibit X;

-24-

- (f) there shall not be in effect any injunction or restraining order issued by a court of competent jurisdiction in an action or proceeding against the consummation of the transactions pursuant to this agreement;
- (g) TISH or TISI shall have duly executed and delivered to Transferor or the Stockholders, as the case may be, the agreements referred to in section 7.1(h);
- (h) the conditions set forth in sections 7.1(i) and 7.1(j) shall be fulfilled; and
- $% \left( 1\right) =0$  (i) the other transactions set forth on schedule 6.11 have been completed.
  - 8. Documents to be Delivered at Closing.
- 8.1 Documents to Be Delivered by Transferor and the Stockholders. At the closing, Transferor and the Stockholders shall deliver to TISH the following:
- (a) such bills of sale, assignments or other instruments of transfer and assignment as shall be effective to vest in TISH title (in

accordance with section 4.4) to the assets being sold and assigned under this agreement;

- (b) a copy of resolutions of the board of directors of Transferor authorizing the execution, delivery and performance of this agreement by Transferor, and a certificate of its secretary or assistant secretary, dated the Closing Date, that such resolutions were duly adopted and are in full force and effect;
  - (c) the certificate referred to in section 7.1(c);
  - (d) the opinion referred to in section 7.1(e); and
- (e) the agreements referred to in sections 2.1(a), 6.8 and 6.9.
- 8.2 Documents To Be Delivered by TISH. At the closing, TISH shall deliver to Transferor and the Stockholders the following:
  - (a) the wire transfer of funds referred to in section 2.1;
- (b) certificates representing the number of shares of Class B Non-Voting Common Stock and Class C Non-Voting Common Stock of TISH referred to in section 2.1.
- (c) instruments pursuant to which TISH assumes the obligations and liabilities to be assumed by it under section 2.1(c);

-25-

- (d) a copy of resolutions of the board of directors of TISH and TOIC authorizing the execution, delivery and performance of this agreement by TISH and TOIC, and a certificate of its secretary or assistant secretary, dated the Closing Date, that such resolutions were duly adopted and are in full force and effect;
  - (e) the certificate referred to in section 7.2(c);
  - (f) the opinion referred to in section 7.2(e); and
- (g) the agreements referred to in section 2.1(a), 6.8 and 6.9.
  - 9. Survival of Representations and Warranties; Indemnification.
    - 9.1 Survival.
- (a) All representations, warranties and agreements by Transferor and the Stockholders shall survive the closing (but only for the respective periods set forth in section 9.3 below) notwithstanding any

investigation at any time by or on behalf of TISH or TOIC, and shall not be considered waived by TISH's or TOIC's consummation of the acquisition contemplated by this agreement with knowledge of any breach or misrepresentation by Transferor and the Stockholders.

(b) All representations, warranties and agreements by TISH and TOIC shall survive the closing (but only for the respective periods set forth in section 9.3 below) notwithstanding any investigation at any time by or on behalf of Transferor and the Stockholders, and shall not be considered waived by Transferor's and the Stockholders' consummation of the acquisition contemplated by this agreement with knowledge of any breach or misrepresentation by TISH or TOIC.

## 9.2 Indemnification; Exclusive Remedy.

(a) Subject to the other sections of this Article 9, Transferor and the Stockholders shall, jointly and severally, indemnify and hold harmless TISH and TOIC against all loss, liability, damage or expense (including reasonable fees and expenses of counsel, whether involving a third party or between the parties to this agreement) TISH or TOIC may suffer, sustain or become subject to as a result of (i) any breach of any warranty of Transferor and the Stockholders contained in this agreement, or any misrepresentation by Transferor and the Stockholders (determined without regard to any materiality, material adverse effect or knowledge qualifiers), or any claim by a third party which, without regard to the merits of the claim, would constitute such a breach or misrepresentation, (ii) any breach of any covenant or other agreement of Transferor or the Stockholders contained in this agreement or any claim by a third party which, without regard to the merits of the claim, would constitute such a breach, (iii) Transferor's or the Stockholders' failure to pay, perform or discharge when due any of Transferor's and the Stockholders' obligations, liabilities, agreements or commitments not expressly assumed by TISH as an Assumed Liability pursuant to this agreement (including, but

-26-

not limited to, any liability with respect to any claim, litigation or proceeding arising out of the operation of the Business prior to the Closing Date which was not disclosed on a schedule to this agreement), (iv) any claims made by MSE or any person claiming to have an interest in the capital stock or assets of Transferor arising out of the transactions contemplated by this agreement (but not any claims relating to any business or other relationship between MSE and TISH or TOIC or any of their affiliates, which are unrelated to the transactions contemplated by this agreement) [, or (v) any claims or liabilities with respect to the Excluded Business].

(b) Subject to the other sections of this Article 9, TISH and TOIC shall, joint and severally, indemnify and hold harmless Transferor and the Stockholders against all loss, liability, damage or expense (including reasonable fees and expenses of counsel, whether involving a third party or between the parties to this agreement) Transferor and the Stockholders may

suffer, sustain or become subject to as a result of (i) any breach of any warranty contained in this agreement or any misrepresentation by TISH or TOIC, or any claim by a third party which, without regard to the merits of the claim, would constitute such a breach or misrepresentation, (ii) any breach of covenant or other agreement contained in this agreement by TISH or TOIC, or any claim by a third party which, without regard to the merits of the claim, would constitute such a breach or (iii) TISH's failure to pay, perform and discharge when due any of the Assumed Liabilities.

(c) Except for the ability to seek injunctive or similar equitable relief to enforce affirmative covenants, the remedy of TISH, TOIC and their affiliates against the Stockholders and Transferor pursuant to this agreement, any documents or instruments pursuant to this agreement, or at law, equity or otherwise, is limited solely to the indemnification provided in this section 9 and is the only remedy of any of them against the Stockholders or Transferor, and TISH and TOIC, for themselves and their affiliates, hereby expressly waive all rights and remedies whatsoever, whether by statute, rule, regulation, in tort or otherwise, against the Stockholders except for the right of indemnification under this section 9.2, provided that the foregoing shall not apply to (x) claims of fraud, (y) the covenants set forth in sections 6.7, 6.12, 6.14 and 6.15 and (z) the employment and consulting agreements. TISH and TOIC, for themselves and their affiliates, acknowledge that (i) neither the Transferor nor the Stockholders have made any representations or warranties other than as expressly set forth in section 4 hereof, (ii) TISH, TOIC and their affiliates have made their own independent examination, investigation, analysis and evaluation of Transferor and the Stockholders and have undertaken such due diligence as they have deemed adequate, which examination, investigation, analysis, evaluation and due diligence shall not limit or diminish in any manner the representations and warranties and obligations of Transferor and the Stockholders under this agreement, and (iii) damages of TISH, TOIC and their affiliates, if any, may be greater than the funds in escrow pursuant to the Escrow Agreement.

(d) Transferor and the Stockholders acknowledge that (i) neither TISH nor TOIC have made any representations or warranties other than as expressly set forth in section 5 hereof, and (ii) Transferor and the Stockholders have made their own independent examination, investigation, analysis and evaluation of TISH and TOIC and have undertaken such due diligence as they have deemed adequate, which examination, investigation, analysis, evaluation and due

-27-

diligence shall not limit or diminish in any manner the representations and warranties and obligations of TISH and TOIC under this agreement.

9.3 Notices of Claims. No party shall be liable for misrepresentation or breach of warranty except to the extent that notice of a claim is given by another party in writing on or prior to April 30, 2001, except that, (a) for misrepresentation or breach of warranty under sections 4.2, 4.4(a)

and 4.15 notice of a claim must be given in writing to Transferor or the Stockholders within ten (10) years after the Closing Date, (b) with respect to misrepresentation or breach of warranty relating to taxes and ERISA, notice of a claim must be given in writing to Transferor or the Stockholders no later than three months after the expiration of the statute of limitation applicable to the tax involved, (c) with respect to misrepresentation or breach of warranty under section 4.13, notice of a claim must be given in writing to Transferor within six years after the Closing Date, and (d) for misrepresentation or breach of warranty under sections 5.2, 5.4 and 5.5(e) notice of a claim must be given in writing to TISH or TOIC within ten (10) years after the Closing Date. Prompt written notice upon assertion of a claim that may be indemnified must be given to the party liable for misrepresentation or breach of warranty, but the failure to provide such notice shall not relieve or otherwise affect the obligation of the party liable to provide indemnification hereunder, except to the extent that any Losses (as hereinafter defined) directly resulted from or were caused by such failure.

9.4 Basket. Neither Transferor or the Stockholders nor TISH shall be liable for misrepresentation or breach of warranty under this agreement unless and until the aggregate amount of losses, liabilities, damages and expenses incurred by an indemnified party ("Losses") as a result of all misrepresentations and breaches of warranty under this agreement, together with all such Losses incurred by such party or its affiliates under the Affiliate Acquisition Agreements, exceeds \$75,000 in the aggregate (in which event it shall be liable for the full amount of all Losses). No claim shall be asserted by a party against another party unless such claim, when aggregated with other claims asserted at the same time, involve at least \$10,000 of claimed Losses. This provision shall not apply to any misrepresentation or breach of warranty under sections 4.4(a), 4.18, 5.5(e) or to claims asserted under sections 9.2(a)(ii), (iii), (iv) or (v) or 9.2(b)(ii) or (iii). In calculating the amount of the loss, liability, damage or exposure to TISH or Transferor for misrepresentation or breach of warranty by the other which is indemnified under section 9.2, all materiality, material adverse effect and knowledge qualifiers set forth in a representation or warranty that has been misrepresented or breached shall be disregarded.

9.5 Cap on Liability. Notwithstanding anything to the contrary in this agreement, the aggregate liability of Transferor, the Stockholders and the Affiliates to TISH for misrepresentation or breach of warranty under this agreement and the Affiliate Acquisition Agreements shall be limited to \$3,000,000 ("Cap"); provided that such Cap shall not apply to breach of sections 4.2, 4.4(a), 4.18 nor to any claim asserted under section 9.2(a)(ii), (iii), (iv) or (v) or to breach of, or any claim asserted under, the comparable provisions of the Affiliate Acquisition Agreements. All claims shall be first asserted against any amounts held in escrow pursuant to the Escrow Agreement (the "Escrow"), and any claims to which the Cap applies may only be asserted against funds held in the Escrow, there being no recourse against the Stockholders directly for such claims, except that in the event that any amounts held in the

Escrow are reduced or reserved against to pay Non-Cap Claims (as defined below), the Stockholders shall be jointly and severally liable for claims to which the Cap applies up to the amount that was paid or reserved against in connection with the Non-Cap Claims. "Non-Cap Claims" shall mean claims for indemnification under this agreement and the Affiliate Acquisition Agreements that are not subject to the Cap.

## 9.6 Third Party Claims.

- (a) If any party entitled to be indemnified pursuant to section 9.2 (an "Indemnified Party") receives notice of the assertion by any third party or any claim or of the commencement by any such third person of any action (any such claim or action, an "Indemnifiable Claim") with respect to which another party hereto (an "Indemnifying Party") is or may be obligated to provide indemnification, the Indemnified Party shall promptly notify the Indemnifying Party in writing (the "Claim Notice") of the Indemnifiable Claim; provided, that the failure to provide such notice shall not relieve or otherwise affect the obligation of the Indemnifying Party to provide indemnification hereunder, except to the extent that any Losses directly resulted from or were caused by such failure.
- (b) The Indemnifying Party shall have thirty (30) days after receipt of the Claim Notice to undertake, conduct and control, through counsel of its own choosing, and at its expense, the settlement or defense thereof, and the Indemnified Party shall cooperate with the Indemnifying Party in connection therewith; provided that (i) the Indemnifying Party shall permit the Indemnified Party to participate in such settlement or defense through counsel chosen by the Indemnified Party (subject to the consent of the Indemnifying Party, which consent shall not be unreasonably withheld), provided that the fees and expenses of such counsel shall not be borne by the Indemnifying Party, and (ii) the Indemnifying Party shall not settle any Indemnifiable Claim without the Indemnified Party's consent (which consent shall not be unreasonably withheld). So long as the Indemnifying Party is vigorously contesting any such Indemnifiable Claim in good faith, the Indemnified Party shall not pay or settle such claim without the Indemnifying Party's consent, which consent shall not be unreasonably withheld.
- (c) If the Indemnifying Party does not notify the Indemnified Party within thirty (30) days after receipt of the Claim Notice that it elects to undertake the defense of the Indemnifiable Claim described therein, the Indemnified Party shall have the right to contest, settle or compromise the Indemnifiable Claim in the exercise of its reasonable discretion; provided, that the Indemnified Party shall notify the Indemnifying Party of any compromise or settlement of any such Indemnifiable Claim.

#### 10. Miscellaneous.

10.1 Entire Agreement. This agreement (including the exhibits and schedules hereto, which shall be deemed an integral part of this agreement) contains, and is intended as, a complete statement of all of the terms of the

parties with respect to those matters (including, but not limited to, the letter of intent dated August 9, 1999), and cannot be changed or terminated orally.

- 10.2 Governing Law. This agreement shall be governed by and construed in accordance with the law of the State of New York applicable to agreements made and to be performed entirely in New York.
- 10.3 Headings. The section headings of this agreement are for reference purposes only and are to be given no effect in the construction or interpretation of this agreement.
- 10.4 Notices. All notices and other communications under this agreement shall be in writing and shall be deemed given when delivered personally or mailed by registered mail, return receipt requested or faxed to the parties at the following addresses (or to such other address as a party may have specified by notice given to the other party pursuant to this provision):

(a)	If to	Transferor	and	the	Stockholders,	addressed	to	them	at:
	Attent								
	Fax Nu	umber:							
	with a	a copy to:							
	Attent	 tion:							
	Fax Nu								

(b) If to TISH or TOIC, addressed to it at:
 The Official Information Company
 250 West 57th Street,

Suite 2421 New York, New York 10019 Attention:

Fax Number: (212) 247-0026

-30-

with a copy to:
Steven L. Kirshenbaum, Esq.
Proskauer Rose LLP
1585 Broadway
New York, New York 10036
Fax Number: (212) 969-2900

- 10.5 Separability. If any provision of this agreement is held to be invalid or unenforceable, the balance of this agreement shall remain in effect.
- 10.6 Waiver. Any party may waive compliance by another with any of the provisions of this agreement. No waiver of any provision shall be construed as a waiver of any other provision. Any waiver must be in writing and must be signed by the party waiving the provision.
- 10.7 Assignment. TISH may assign its rights under this agreement simultaneously with or after the closing to any subsidiary or affiliate of TISH that has assumed (in a writing reasonably satisfactory in form and substance to Transferor or the Stockholders) TISH's obligations under this agreement, but no such assignment shall relieve TISH of its obligations under this agreement. TISH may designate any subsidiary or other affiliate to receive the Assets and assume the Assumed Liabilities at the closing hereunder; provided that no such designation shall relieve TISH of its obligations under this agreement. In addition, TISH, or any subsidiary or affiliate of TISH to whom this agreement has been assigned in whole or in part, may assign its rights under this agreement to its lenders. This agreement will be binding upon, inure to the benefit of and be enforceable by the parties and their respective successors and permitted assigns.

### 10.8 Arbitration.

(a) Any dispute or controversy arising under or in connection with this agreement other than under the provisions of section 2.3(b), shall be settled exclusively by arbitration to be held in the City of Chicago in accordance with the rules of the American Arbitration Association then in effect; provided that any claim of breach with regard to section 6.7 may, at the option of TISH, be enforced in court. The arbitral tribunal shall be composed of three arbitrators, one of which shall be appointed by TISH, one of which shall be appointed by Transferor and one of which shall be appointed by the arbitrators appointed by TISH and Transferor. As part of their award, the arbitrators shall make a fair allocation between the parties of the fee and expenses of the American Arbitration Association and the cost of any transcript,

taking into account the merits of their claims and defenses. Either party shall be entitled to the pre-trial discovery it would have been entitled to if the proceeding had been brought in a New York court under the Civil Practice Law and Rules. Judgment may be entered on the arbitrators' award in any court having jurisdiction, and the parties irrevocably consent to the jurisdiction of the Chicago courts for that purpose. The parties waive personal service in connection with any such arbitration; any process or other papers under this provision may be served by registered mail, return receipt requested, or by personal service, provided a reasonable time for appearance or response is allowed. The arbitrators may grant injunctive or other relief.

-31-

(b) The courts of the State of Illinois in Cook County and the United States District Court for the Northern District of Illinois shall have jurisdiction over the parties with respect to any dispute or controversy arising under or in connection with section 6.7 and, by execution and delivery of this agreement, each of the parties to this agreement submits to the jurisdiction of those courts, including but not limited to, the in personam and subject matter jurisdiction of those courts, waives any objection to such jurisdiction on the grounds of venue or forum non conveniens, the absence of in personam or subject matter jurisdiction and any similar grounds, consents to service of process by mail (in accordance with section 10.4) or any other manner permitted by law, and irrevocably agrees to be bound by any judgment rendered thereby in connection with section 6.7. These consents to jurisdiction shall not be deemed to confer rights on any person other than the parties to this agreement.

- 10.9 No Third Party Beneficiaries. This agreement does not create, and shall not be construed as creating, any rights in favor of any person not a party to this agreement.
- 10.10 Publicity. Neither TISH nor Transferor, nor their respective affiliates, shall issue or cause the publication of any press release or other public announcement with respect to the transactions contemplated by this agreement without the consent of the other party, which consent shall not be unreasonably withheld or delayed, except as may be required by law or the regulations or policies of any securities exchange, in which case the party required to make the release or statement shall allow the other party reasonable time to comment on such release or statement in advance of such issuance.
- 10.11 Guarantee of TOIC. TOIC hereby guarantees the full and timely performance of all covenants and obligations of TISH hereunder.
- 10.12 Counterparts. This agreement may be executed in counterparts, which together shall constitute the same instrument.

TISI HOLDINGS, INC.

By:

	 Name: Title:		
THE	OFFICIAL	INFORMATION	COMPANY
By:			
	Name: Title:		
[Tr	ansferor]		
-32-			
By:			
	Name: Title:		
[St	ockholder]		
 [St	ockholder]		
-33-			

### STOCKHOLDERS AGREEMENT

## March 15, 2000

The parties to this agreement ("Agreement") are TISI Holdings, Inc., a Delaware corporation (the "Company"), which is a wholly-owned subsidiary of TOIC (hereafter defined), The Official Information Company, a Delaware corporation ("TOIC"), and the parties listed on exhibit I (collectively, the "Stockholders").

The Stockholders, the Company and TOIC have entered into certain Asset Acquisition Agreements and Stock Acquisition Agreements dated March 15, 2000 pursuant to which the assets or the stock of the Stockholders or certain companies owned by the Stockholders, as the case may be, listed on exhibit II are being transferred to the Company for cash and an aggregate of 2,400 shares of Class B Non-Voting Common Stock of the Company ("Class B Stock") and 5,900 shares of Class C Non-Voting Common Stock of the Company ("Class C Stock") representing 4.477% of the outstanding equity of the Company on the date hereof. TOIC has transferred to the Company all of the outstanding shares of its wholly-owned subsidiary, Total Information Services, Inc., an Oklahoma corporation, in exchange for a total of 177,102 shares of Class A Voting Common Stock of the Company ("Class A Stock") constituting all the issued and outstanding shares of Class A Stock and representing 95.523% of the outstanding equity of the Company on the date hereof.

The parties agree as follows:

- 1. Management of the Company.
- 1.1 Board of Directors. The business and affairs of the Company shall be managed under the direction of a board of directors consisting of 3 members or such other number as the board or the holders of the Class A Stock may from time to time determine. The board of directors shall be elected exclusively by the holders of the Class A Stock.
- 1.2 Advisory Committee. The Company shall establish an advisory committee consisting of six members, three of whom shall be designated by the Stockholders (by a vote of the holders of a majority of the Class B Stock and Class C Stock voting together) and the remaining three of whom shall be designated by TOIC. TOIC shall designate the chairman of the advisory committee. The advisory committee shall meet at least four times during the first year of this agreement and semi-annually thereafter, on such dates as shall be determined by the chairman, to review the progress of the Company, discuss the Company's strategic focus and review material business proposals that are not in the ordinary course of business, which are being presented for consideration by the board of directors, including any proposals relating to acquisitions or dispositions by the Company. The advisory committee shall periodically report

its views to the board of directors but does not have authority to bind the Company or the board of directors in connection with any matter.

1.3 Voting. Holders of Class B Stock and Class C Stock shall have no voting rights, except to the extent required by law.

## 2. Transfers of Interests.

- 2.1 Restrictions on Transfers. Except with the prior written consent of TOIC, which consent may be withheld by TOIC in its sole discretion, or except as provided in Sections 2.3, 2.4 or 2.6, no Stockholder may sell, transfer, assign, pledge, grant a security interest in or otherwise dispose of or encumber ("transfer") all or any portion of his, her or its shares of Class B Stock or Class C Stock, and any purported transfer shall be void. Any transfer by TOIC of its shares of Class A Stock shall be subject to the terms and conditions of Sections 2.2, 2.3, 2.4 and 2.5.
- 2.2 Conditions to Transfer. No transfer of an interest in the Company shall be effective unless the transferee (a) agrees in writing to be bound by the terms of this Agreement as if the transferee were the transferor and (b) executes such other documents and agreements as the board of directors reasonably may request.
- 2.3 Drag Along. If at any time TOIC shall propose to sell any of its shares of Class A Stock (or any other class of common stock acquired pursuant to Section 7) to an unrelated third party that is willing to purchase such shares of capital stock in the Company, the Stockholders also shall at the request of TOIC sell a proportionate number of their shares of Class B Stock and Class C Stock (and any other class of common stock acquired pursuant to Section 7) to the third party at the same price per share and on the same terms and conditions as TOIC. TOIC shall give notice (specifying the identity of the prospective purchaser, the proposed purchase price, the scheduled date of the closing, and all other relevant material information) to all of the Stockholders at least 10 business days prior to the closing of the sale, and the sale by all of the stockholders of the Company shall take place simultaneously.
- A Stock (or any other class of common stock acquired pursuant to Section 7) to an unrelated third party, TOIC shall so notify each of the Stockholders (specifying the identity of the prospective purchaser, the proposed purchase price, the scheduled date of the closing, and all other relevant material information), and each of the Stockholders may elect, by notice to TOIC given within 10 business days from the date of the notice from TOIC, to sell up to the same percentage of its shares of Class B Stock or Class C Stock (or any other class of common stock acquired pursuant to Section 7), as the case may be, that TOIC proposes to sell of its Class A Stock (or any other class of common stock) to such third party at the same purchase price and on the same terms and conditions as the sale by TOIC. If any Stockholder does not elect to sell the full pro rata portion of its shares in the Company that it is permitted to sell pursuant to the preceding sentence, TOIC and each of the Stockholders who

stockholder's percentage ownership of capital stock in the Company to cover the amount that the Stockholders who did not elect to sell could have sold. If any Stockholder elects to sell any portion of his, her or its shares of Class B Stock or Class C Stock (or any other class of common stock) then, simultaneously with the sale, TOIC shall cause the proposed purchaser to purchase from each of the electing Stockholders, and each of the electing Stockholders shall sell to the proposed purchaser, the number of shares that such Stockholder is eligible to sell as determined above, and the net proceeds of such sales shall be allocated proportionately among the selling stockholders.

- 2.5 Purchase Price. The purchase price for the shares of Class B Stock and Class C Stock (and any other class of common stock) of the Stockholders under Sections 2.3 and 2.4 shall include the pro rata portion of any consideration payable to TOIC or its affiliates (including any noncompetition payments and the like) by an unrelated third party. If any consideration is in the form of property other than cash or cash equivalents, the fair market value thereof shall be determined by the board of directors in good faith. Under no circumstances shall there by any differentiation in the per share purchase price among any Class A Stock, Class B Stock, Class C Stock or any other class of common stock.
- 2.6 Transfers to Affiliates; Family Members. Notwithstanding the provisions of this Section 2 or any other provisions to the contrary contained in this Agreement, (i) a Stockholder that is a corporation may, at any time or from time to time, transfer all or any portion of its shares of Class B Stock and Class C Stock to its stockholders and (ii) a Stockholder who is a natural person may transfer all or any portion of his or her shares of Class B Stock and Class C Stock to his or her spouse or his or her lineal descendants or to a trust of which they are the sole beneficiaries, or, upon his or her death, have his or her shares of Class B Stock and Class C Stock transferred to his or her estate; provided, however, the transferee in any of the above cases shall execute an agreement to be bound by the terms and conditions of this Agreement in form reasonably acceptable to the Company.
- 2.7 Initial Public Offering. The "drag along" and "tag along" rights contained in Sections 2.3 and 2.4 shall not apply to any sales by TOIC in connection with an initial public offering of the Company's shares or other interests of a successor, and all of the restrictions contained in Section 2 shall terminate upon consummation of such an initial public offering, except all shares shall continue to be subject to requirements under the securities laws of the United States.
- 2.8 Ownership of Other Forms of Capital Stock. In the event that TOIC and one or more Stockholders receives a different form of capital stock pursuant to Section 7, and TOIC proposes to sell such other form of capital

stock, TOIC and the Stockholders that have acquired such other form of capital stock shall have the "drag along" and "tag along" rights described in Sections 2.3 and 2.4 with respect to such securities.

3

- 3. Public Offering; Registration Rights.
- 3.1 Conversion. In the event of an initial public offering by the Company, or upon an earlier decision by the board of directors, all shares of Class A Stock, Class B Stock and Class C Stock shall be converted into the same class of voting common stock on a one-for-one basis (such shares of common stock owned by TOIC and the Stockholders herein called the "Shares").
- 3.2 "Piggy Back" Registration Rights. If at any time in or after the consummation of an initial public offering by the Company or a successor (referred to in this Section 3 as the "Company"), the Company shall elect to file a registration statement under the Securities Act of 1933, as amended (the "Act"), for the account of TOIC (other than a registration statement relating solely to shares issuable to employees under stock option or other employee benefit plans), the Company shall give prompt notice to each of the Stockholders and/or their respective assigns or successor(s) in interest (referred to collectively and in the alternative in this Section as "Holders") of its intention to effect such a registration. Upon the request of any Holder, the Company shall include in the registration statement the Shares specified by such Holder in such notice(s), except that the Company shall not be obligated to include the Shares in the registration statement if either of the following conditions exists:
- (a) The managing underwriter or underwriters advise the Company in writing that in its or their opinion, or in the case of a registration not being underwritten, the Company shall reasonably determine after consultation with an investment banker of nationally recognized standing, that the number of securities proposed to be sold pursuant to such offering exceeds the number of securities which can be effectively sold in, or would have a material adverse effect on, such offering. In such event, the Company will include in such registration the number of securities which, in the opinion of such managing underwriter or underwriters, or the Company, as the case may be, can be sold as follows: (i) first, the securities the Company proposes to sell, if any, and (ii) second, the securities requested by TOIC, the Holders and other stockholders, if any, of the Company that have shares to be included in such registration. If the Company includes some but not all of the securities requested by TOIC, the Holders and the other stockholders to be registered, the Company shall include the pro rata portion of the requested amount of each party.
- (b) Holders giving such notice holding not less than a majority of the Shares to be registered under the registration statement shall have failed to agree in writing within 10 days after the Company's request to do so as follows: (i) not to sell any of their shares for at least 180 days after the

effective date of the registration statement and thereafter to sell the shares so registered from time to time only in such amounts as the Company may reasonably prescribe for purposes of preserving an orderly market in the Company's shares, or (ii) to distribute the shares for which registration was requested (or such lesser number of shares, in proportion to the total number of shares to be offered pursuant to the registration statement, as the managing underwriter or underwriters may specify) pursuant to a firm underwriting through a managing underwriter or underwriters designated by the Company. If the Company requests that

4

the Holders agree to the restrictions of this Section 3.2(b), TOIC shall also be required to agree to such restrictions.

- 3.3 Agreements of the Stockholders. TOIC and the Stockholders agree that in the event of an offering under Section 3.2, (i) for a period of 180 days after the date of the registration statement, they shall not, without the prior written consent of the Company or any underwriter, offer, issue, sell, contract to sell, grant any option for the sale of, or otherwise dispose of, directly or indirectly, any Shares and (ii) TOIC and the Holders of any Shares included in a registration statement shall (a) furnish to the Company and the underwriter of such offering, if any, such information and other material and execute such documents as may reasonably be requested, (b) make any requested representations and warranties in connection with the accuracy of information provided with respect to TOIC or such Holders, as the case may be, (c) enter into any further agreements as the Company or any underwriter may reasonably request and (d) provide such indemnities to the Company and any underwriter as may be reasonably requested.
- 3.4 Expenses. The Company shall bear all of the costs and expenses of the registration to which this Section 3 applies except for the costs and expenses of Holders comprising broker's and underwriter's commissions or fees, transfer taxes and the costs and expenses of counsel, accountants or other advisers engaged by the Holders which shall be borne by the Holders in the same proportion as the number of shares to be offered for the account of any Holder bears to the total number of shares being offered by all of the Holders.

## 4. Put/Call.

- 4.1 Sale of Shares. If an initial public offering of the Company's securities has not been consummated within three years of the date of this Agreement, each Stockholder may elect, by notice to the Company, given within three months after the third anniversary of this Agreement, to sell to the Company all or a portion of the shares of Class B Stock held by such Stockholder.
- 4.2 Purchase of Shares. If an initial public offering of the Company's securities has not been consummated within three years of the date of this Agreement, the Company may elect, by notice to the Stockholders, given

within three months after the third anniversary of this Agreement, to purchase all or a portion of the shares of Class B Stock held by the Stockholders.

- 4.3 Purchase Price. The purchase price of any shares of Class B Stock purchased by the Company pursuant to (i) Section 4.1 shall be equal to \$1,000 per share and (ii) Section 4.2 shall be equal to \$2,000 per share.
- 4.4 Closing. The closing of any sale and purchase pursuant to Section 4.1 or 4.2 shall be held at the Company's principal executive offices on a date set by the Company, by notice to the Stockholders, not later than 30 days after the date of the notice of election to sell or purchase. At the closing, the Stockholders shall transfer to the Company the

5

shares of Class B Stock to be sold, and the Company shall pay the purchase price for the shares by wire transfer or certified or bank cashier's check.

- 5. Debt, Dividends and Related Distributions.
- (a) The Stockholders and TOIC acknowledge that the total enterprise value of the Company on the date of this Agreement is \$241,992,000, \$185,402,000 of which shall be allocated as equity and \$56,590,000 of which shall be allocated as debt payable to TOIC from the Company ("Attributable Debt"). The Attributable Debt shall bear interest at a rate of 10-3/8% per year and shall be due in 2007, subject to earlier repayment in accordance with Section 5(b).
- (b) All Free Cash Flow (as defined below) of the Company shall be applied periodically to first pay accrued interest on all Attributable Debt and then, if no amount remains accrued and unpaid thereunder, to repay the principal of such Attributable Debt. To the extent that the Attributable Debt is fully repaid, any additional Free Cash Flow shall be loaned periodically by the Company to TOIC (the "Company Loan"). The Company Loan shall bear interest, compounded quarterly, at the same rate as is payable from time to time by TOIC on its secured senior debt, which interest shall accrue and be payable, together with the outstanding amount of the Company Loan, upon a sale of all or substantially all of the assets of the Company or TOIC or all or substantially all of the capital stock in the Company or TOIC.
- (c) The Stockholders acknowledge that the Company may incur future indebtedness to TOIC for general corporate purposes, including working capital, capital expenditures and acquisitions, upon the same terms as the Attributable Debt. In the event that future indebtedness is incurred through an acquisition by the Company, such debt shall not exceed the lesser of (i) the product of (a) the pro forma EBITDA of the Company, after giving effect to the acquisition, for the 12 months prior to the acquisition and (b) 3.75 or (ii) an amount based on market-rate lending limits for companies of equivalent size and comparable nature, as determined by the board of directors in good faith.

- (d) "Free Cash Flow" means the net income of the Company plus amortization and depreciation plus (or minus) changes in working capital (less) capital expenditures plus the after tax proceeds of any divestiture of assets (net of costs of the transaction) less any cash used to make an acquisition (including costs incurred with regard to such acquisition), as determined in good faith by the board of directors of the Company.
- (e) Shares of Class A Stock, Class B Stock and Class C Stock shall have identical rights with respect to liquidation preference and dividends and other distributions.
  - 6. Change in Control of TOIC.
- 6.1 Sale by the Stockholders. The Stockholders shall be entitled, and if requested by TOIC, required, to sell to TOIC or its designee, all of their capital stock upon a

6

Change in Control (as defined in Section 6.2) of TOIC. The purchase price for such capital stock shall be determined as follows (the "Agreed Determination Method"):

- (a) by mutual agreement between the Stockholders and TOIC; or
- (b) if the Stockholders and TOIC are unable to agree upon a valuation, by an independent, nationally recognized investment bank, designated by agreement between the Stockholders holding a majority of the shares of Class B Stock and TOIC, with experience in transactions with, and valuations of companies in, the same business as the Company, whose decisions on the matter shall be binding and whose fees and expenses shall be borne 25% by the Stockholders and 75% by TOIC.
- 6.2 Definition. The term "Change in Control" means (A) a sale by the stockholders of TOIC of 50% or more of the outstanding capital stock of TOIC, (B) a sale of all or substantially all of the assets of TOIC, which assets include the stock of the Company or (C) a merger or consolidation involving TOIC in which TOIC is not the surviving entity, except a merger undertaken solely to change jurisdictions, or following which the stockholders of TOIC immediately preceding such merger or consolidation, as a group own directly or indirectly more than 50% of the equity or voting interests in the surviving entity.
- 6.3 Procedure. TOIC shall give notice to the Stockholders of a Change in Control at least 10 business days prior to the anticipated closing of such a transaction and of whether TOIC has elected to require the Stockholders to sell their shares of capital stock. If TOIC does not elect to require the Stockholders to sell their shares, the Stockholders shall have 10 business days from receipt of such notice to notify TOIC of their intention to sell such shares. Payment pursuant to this section 6 shall be made to the Stockholders upon the consummation of the Change of Control transaction.

7. Future Equity Investments. In the event that the board of directors of the Company determines that additional funds are necessary to maintain any of the Company's business activities, to avoid or cure any default (to the extent not waived by the appropriate lender) under any agreement or instrument relating to indebtedness for borrowed money, for the acquisition of any other entity or business or for any other purpose, the Company may request additional capital contributions from TOIC and the Stockholders, and TOIC and each Stockholder may (but shall not be obligated to) contribute to the Company its pro-rata share of the cash to be contributed based on TOIC's and such Stockholder's percentage ownership of capital stock in the Company. Any such request for additional capital contributions shall set forth (a) the aggregate amount of capital requested from TOIC and the Stockholders and the proposed use of that capital; (b) the proportionate share requested from TOIC and each Stockholder; and (c) the due date for, and the method and place of, payment of the contribution (which shall be the same for TOIC and each Stockholder). If TOIC or any Stockholder does not make its entire additional capital contribution by the due date, the Company shall give each of TOIC and the other Stockholders an opportunity to make additional contributions in any amount up to the amount required (and if TOIC and the other Stockholders are willing to contribute more than the amount required, TOIC and such Stockholders shall make the additional contribution in

7

proportion to the additional amounts each was willing to contribute). Only shares of Class A Stock shall be issued to TOIC and shares of Class C Stock shall be issued to the Stockholders in connection with this Section 7, the number of which shall be determined by the Agreed Determination Method (with all shares of such stock, regardless of class, being valued at the same price). Notwithstanding anything in this Section 7 to the contrary, the Company may fund the various business activities described in this Section 7 through the issuance of equity securities in the Company, including equity securities of different classes and forms than those outstanding on the date hereof ("Other Securities"). In the event of the issuance by the Company of Other Securities in connection with the funding of business activities, TOIC and the Stockholders, to the extent that they have made capital contributions in connection with such funding, shall acquire the same class and form of equity securities as issued by the Company, except that the Stockholders shall only be entitled to acquire non-voting equity securities. The number of shares of Other Securities acquired by TOIC and the Stockholders in connection with this Section 7 shall be determined by the Agreed Determination Method (with all shares of common stock, regardless of class, being valued at the same price). Notwithstanding anything in this Section 7 to the contrary, the Company may issue shares of its capital stock in connection with an acquisition of any business or entity without providing TOIC or the Stockholders the opportunity to purchase any additional shares or exercise preemptive rights.

8. Noncompete. So long as the Stockholders own shares of capital stock of the Company, TOIC shall not, directly or indirectly, engage, or be interested

in (as owner, stockholder, partner, member, manager, lender, agent or otherwise) or provide any service to any business or entity that engages, anywhere in the United States, in the operation of pre-employment investigation and screening services, except through the Company or an entity controlled by the Company.

- 9. Representations and Warranties. Each of the parties hereto severally, as to itself, himself or herself, and not jointly, hereby represents and warrants to each of the other parties to this Agreement that:
  - (i) such party has the full right, power and authority to execute, deliver and perform this Agreement;
  - (ii) this Agreement has been duly executed and delivered by or on behalf of such party and constitutes a legal, valid and binding obligation of such party, enforceable against such party in accordance with its terms;
  - (iii) no consent, approval, authorization or order of any person is required for the execution, delivery or performance of this Agreement by such party;
  - (iv) neither the execution, delivery nor performance of this Agreement by such party will (A) conflict with, or result in a breach of, or constitute a default under, or result in a violation of, any agreement or instrument to which such party is a party or by which such party or its, his or her property is bound, or (B) result in the violation of

8

any applicable law or order, judgment, writ, injunction, decree or award of any governmental authority; and

(v) such party does not have a present plan or intention to sell, assign, transfer or otherwise dispose of the shares of capital stock of the Company (or any interest therein) held by each such party.

Each of the parties hereto agrees that the representations and warranties set forth in this Section 8 shall survive the execution and delivery of this Agreement.

# 10. Miscellaneous

10.1 Legend. Each certificate evidencing the shares of Class A Stock, Class B Stock and Class C Stock shall include a legend in substantially the following form (in addition to any other statements or legends required by law):

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR APPLICABLE STATE SECURITIES LAWS. THESE SECURITIES HAVE BEEN ACQUIRED FOR INVESTMENT AND NOT WITH A VIEW TO DISTRIBUTION OR RESALE, AND MAY NOT BE SOLD, MORTGAGED, PLEDGED, OR

OTHERWISE TRANSFERRED WITHOUT AN EFFECTIVE REGISTRATION STATEMENT FOR SUCH SECURITIES UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND ANY APPLICABLE STATE SECURITIES LAWS. THIS CERTIFICATE AND THE SECURITIES REPRESENTED HEREBY ARE TRANSFERABLE ONLY UPON COMPLIANCE WITH THE PROVISIONS OF THAT CERTAIN STOCKHOLDERS AGREEMENT DATED AS OF MARCH 15, 2000 BY AND AMONG TISI HOLDINGS, INC., THE OFFICIAL INFORMATION COMPANY AND EACH OF THE STOCKHOLDERS NAMED THEREIN, A COPY OF WHICH IS ON FILE IN THE OFFICE OF THE SECRETARY OF TISI HOLDINGS, INC. AND WILL BE MADE AVAILABLE UPON REQUEST TO ANY STOCKHOLDER WITHOUT CHARGE."

10.2 Governing Law. This Agreement shall be governed by and construed in accordance with the law of the state of New York applicable to agreements made and to be performed entirely in New York.

10.3 Notices. All notices and other communications under this Agreement shall be in writing and may be given by any of the following methods: (a) personal delivery; (b) facsimile transmission; (c) registered or certified mail, postage prepaid, return receipt requested; or (d) overnight delivery service. Notices shall be sent to the appropriate party at its, his or her address or facsimile number given below or, with respect to the Stockholders, at the addresses or facsimile numbers as set forth in the records of the Company:

9

if to the Company or TOIC, to each of them at:

250 West 57th Street, Suite 2421 New York, New York 10019 Attention: Brian A. Meyer Fax: (212) 247-0026

with a copy to:

Proskauer Rose LLP 1585 Broadway New York, New York 10036-8299 Attention: Steven L. Kirshenbaum, Esq. Fax: (212) 969-2900

if to the Stockholders, to each of them at the address set forth in the Company's records, with a copy to:

Jenner & Block One IBM Plaza Chicago, Illinois 60611 Attention: Craig R. Culbertson, Esq. Fax: (312) 527-0484

All such notices and communications shall be deemed received upon (a) actual

receipt by the addressee, (b) actual delivery to the appropriate address or (c) in the case of a facsimile transmission, upon transmission by the sender and issuance by the transmitting machine of a confirmation slip confirming the number of pages constituting the notice have been transmitted without error. In the case of notices sent by facsimile transmission, the sender shall contemporaneously mail a copy of the notice to the addressee at the address provided above by any of the methods of delivery specified in clauses (a), (c) or (d) of the paragraph above; however, that mailing shall not alter the time at which the facsimile notice is deemed received.

- 10.4 Counterparts. This Agreement may be executed in counterparts, each of which shall be considered an original, but all of which together shall constitute the same instrument.
- 10.5 Equitable Relief. The parties acknowledge that the remedy at law for breach of this Agreement may be inadequate and that, in addition to any other remedy a party may have for a breach of this Agreement, that party may be entitled to an injunction restraining any such breach or threatened breach, or a decree of specific performance, without posting any bond or security. The remedy in this Section 10.5 is in addition to, and not in lieu of, any other rights or remedies a party may have.

10

- 10.6 Separability. If any provision of this Agreement is invalid or unenforceable, the balance of this Agreement shall remain in effect, and, if any provision is inapplicable to any person or circumstance, it shall nevertheless remain applicable to all other persons and circumstances.
- 10.7 Entire Agreement. This Agreement contains a complete statement of all the arrangements among the parties with respect to its subject matter, supersedes all existing agreements among them with respect to that subject matter, may not be changed or terminated orally and any amendment or modification must be in writing and signed by the party to be charged.

11

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first above written.

TISI HOLDINGS, INC.

By:

Name:
Title:

THE OFFICIAL INFORMATION COMPANY

By:

Name: Title:	
Benjie Walke:	r
Donna Walker	
David A. Sch	lack
William J. Ha	addad
John F. Powe:	rs
Lynda K. Luca	 as
Dorthy L. Nee	el
Debbie D. Arg	do 
Jeff A. Potts	 5
M. Kathleen I	Edwards

Leif J. Lauritzen

Janell A	. Davis
PENJERDE	L MUTUAL ASSOCIATION, INC.
By:	
Name: Title	
	LLC D/B/A UPPER MIDWEST
By:	
Name: Title	
13	
	ON TECHNOLOGY INC. D/B/ARS MUTUAL ASSOCIATION WEST
Name: Title	
EXECUTIV	TE MANAGEMENT SERVICES, INC.
By:	
Name: Title	
NATIONAL	LOSS PREVENTION BUREAU, LTD.
By:	
Name: Title	
NATIONAL GROUP, I	LOSS PREVENTION BUREAU
By:	
Name:	

Title: